QUALIFICATION OF ENGINEERING PROCUREMENT AND CONSTRUCTION (EPC) CONTRACTOR

FOR

UNDERTAKING BALANCE WORKS FOR COMPLETION OF THE PROJECT "ESTABLISHMENT OF CENTRE OF EXCELLENCE IN FARM MECHANISATION"

RWANDA

REFERENCE NO: GOILOC-201(P-4)/Rwanda/EPC-103

Date: December 5, 2025

Funded by:

Export-Import Bank of India



DISCLAIMER

The information contained in this RFQ Document or subsequently provided to the Bidder(s), in documentary or digital form, by or on behalf of Exim Bank, is provided on the terms and conditions set out in this RFQ Document and such other terms and conditions subject to which such information is provided.

This RFQ Document is not an agreement and is neither an offer nor an invitation by Exim Bank to the prospective Bidders. The purpose of this RFQ Document is to provide interested parties with information that may be useful to them in the formulation of their Techno-Commercial Bid for Qualification pursuant to this RFQ Document. The RFQ Document contains statements which reflects various assumptions, assessments and such statements in this RFQ Document do not purport to contain all the information that each Bidder may require. This RFQ Document may not be appropriate for all persons and Bidders and it is not possible for Exim Bank, its employees or advisors to consider the objectives, financial situation and particular needs of each party or Bidder who reads or uses this RFQ Document. The assumptions, assessments, statements and information contained in this RFQ Document may not be accurate, complete or adequate. Each Bidder should, therefore, conduct its own investigations and analysis and should check the accuracy, adequacy, correctness, reliability and completeness of the assumptions, assessments, statements and information contained in this RFQ Document and obtain independent advice from appropriate sources. Exim Bank shall not be liable to any Bidder for any consequential or indirect damages, including but not limited to financial loss which may arise due to any contingent decisions of the Bidder based on this RFQ Document.

Information provided in this RFQ Document to the Bidders is on a wide range of matters, some of which may depend upon interpretation of law. Exim Bank has the right to interpret the submitted Techno-Commercial Bid based on common usage of terminologies and phrases in public procurement. The interpretation of Exim Bank shall be final and binding on the Bidders. The information given is not intended to be an exhaustive account of statutory requirements and should not be regarded as a complete or authoritative statement of law. However, the Bidders must seek appropriate legal advice while formulating the submissions and assess the various risks accordingly.

Exim Bank, its employees and advisors make no representation or warranty and shall have no liability to any person, including any Bidder, under any law, statute, rules or regulations or tort, principles of restitution or unjust enrichment or otherwise for any loss, damages, cost or expense which may arise from or be incurred or suffered on account of anything contained in this RFQ Document or otherwise, including the accuracy, adequacy, correctness, completeness or reliability of the RFQ Document or any assessment, assumption, statement or information contained therein or deemed to form part of this RFQ Document or arising in any way with participation in this process. Exim Bank shall not be liable in any manner whatsoever for any costs or other expenses incurred by a Bidder in preparation or submission of the Techno-Commercial Bid, regardless of the conduct or outcome of the RFQ process.

Exim Bank also accepts no liability of any nature whether resulting from negligence or otherwise howsoever caused arising from reliance of any Bidder upon the statements contained in this RFQ Document. Exim Bank may, at its absolute discretion but without being under any obligation to do so, update, amend or supplement the information, assessment or assumptions contained in this RFQ Document. However, in such a case, appropriate addendum to the RFQ Document shall be issued by Exim Bank.

Selection of a Bidder under this RFQ Document does not entail final selection of the said Bidder as a contractor for the Project. Exim Bank shall not be liable to any Bidder for any consequential or indirect damages, including but not limited to financial loss which may arise due to any contingent decisions of the Bidder based on selection under the terms and conditions contained herein this RFQ Document. The final selection of the contractor for the Project shall be made by the Authority of the borrower country availing Lines of Credit from Government of India. The final contract shall be signed between the Authority and the selected contractor from amongst the Qualified Bidders. Nothing in this RFQ Document shall be construed as creating any contractual relationship between Exim Bank and any Bidder or Qualified Bidder.

The Bidder shall bear all its costs associated with or relating to the preparation and submission of its submittals including but not limited to preparation, copying, postage, delivery fees, expenses associated with any meeting, site visit, demonstrations or presentations which may be required by Exim Bank or any other costs incurred in connection with or relating to its submissions. All such costs and expenses shall be borne by the Bidders and Exim Bank shall not be liable in any manner whatsoever for

the same or for any other costs or other expenses incurred by a Bidder in preparation or submissions, regardless of the conduct or outcome of the Qualification Process.

The Bidder acknowledges that Exim Bank shall not be liable for omission, mistake or error in respect of any of the above or on account of any matter or thing arising out of or concerning or relating to the RFQ Document, including any error or mistake therein.

NOTICE INVITING BIDS

RFQ Reference No.: GOILOC-201(P-4)/Rwanda/EPC-103

Date: December 5, 2025

<u>Qualification of EPC Contractor for completion of Balance Works for Establishment</u> of Centre of Excellence in Farm Mechanization

- 1. The Export-Import Bank of India [hereafter "Exim Bank", or the "Bank"], on behalf of the Government of India ("GOI"), extends from time-to-time, concessional credit facilities under the Indian Development and Economic Assistance Scheme (IDEAS) to overseas governments / their nominated agencies in developing countries for financing projects involving procurement of goods, Works, consultancy and non-consultancy services to be contracted to Indian entities.
- 2. The Lines of Credit (LOCs) are governed by the Guidelines issued by the GOI from time-to-time. As per the extant guidelines issued vide letter F.No. 5/7/2019-IDEAS dated March 31, 2022, Exim Bank is required to carry out a Qualification exercise and provide a list of Qualified Bidders to the Borrower for invitation of bids through a competitive bidding process, for specific contract packages under the project(s) financed by the LOCs.
- 3. In pursuance to the above, Exim Bank invites Bids from eligible Indian entities for Qualification of EPC Contractors for the captioned project.
- 4. The RFQ Document is available for download on Exim Bank's Website. The RFQ Document can be accessed through the following link:

https://www.eximbankindia.in/lines-of-credit-tenders.aspx

- 5. The Bids shall be accepted in an online form in accordance with the provisions and terms and conditions contained herein this RFQ Document. Bid through any other mode shall not be entertained.
- 6. Last date for submission of Bid is January 5, 2026; 15:00 hours.

Mr. Rohan Sharma
Deputy General Manager
Export-Import Bank of India
Infrastructure Group
Office Block, Tower 1
7th Floor, Adjacent Ring Road
Kidwai Nagar [East]
New Delhi -110023

E-mail: <u>infragroup@eximbankindia.in</u> Website: www.eximbankindia.in

SUMMARY OF CONTENT

Part-I: Scope of Works

This Part includes a summary of the prospective project, scope of the works to be undertaken on EPC basis for which this RFQ Process is being conducted.

Part-II: Instructions to Bidders (ITB)

This part provides relevant information to help the Bidders in preparation of Bids in accordance with the RFQ Document. Information is also provided on the submission, opening and evaluation of Bids. This part also specifies the criteria for Qualification of Bidders.

Part-III: Appendix to Instructions to Bidders (AITB)

This part includes provisions specific to each Project for which the Bids are invited through the RFQ Document and supplements "Part-II: Instructions to Bidders (ITB)".

Part-IV: Forms and Appendices

This Section includes the forms and formats of authorization, required documents to be submitted by the Bidder as a part of the Bid.

TABLE OF CONTENT

DISCLAIMER	1
NOTICE INVITING BIDS	4
SUMMARY OF CONTENT	5
TABLE OF CONTENT	6
PART-I: SCOPE OF WORKS	9
PART-II: INSTRUCTIONS TO BIDDERS (ITB)	20
SECTION-1 INTRODUCTION	20
1.1 Background	20
1.2 Brief description of RFQ Process	20
1.3 Schedule of RFQ Process	22
1.4 Reporting Currency	22
1.5 Contents of the RFQ Document	23
1.6 Correspondence with Exim Bank	23
SECTION-2 ELIGIBILITY CRITERIA AND REQURIEMENTS	24
2.1 Bidders	24
2.2 Bidder's Nationality and Ownership	24
2.3 Conflict of Interest	25
2.4 Financial Conduct	
2.5 Debarment / Blacklisting or Conviction	27
2.6 Past Contract Non-Performance	
2.7 Pending Contract Litigation	29
2.8 Contract Litigation History	
2.9 Financial Status and Capacity	
2.10 Bid Capacity	
2.11 General Experience	31
2.12 Specific Experience	33
2.13 Quality Assurance	
2.14 Contractor's Equipment	
2.15 Skilled Human Resource	
2.16 Manufacturer's Authorisation	36
2.18 Additional Requirement	
2.19 Additional Conditions for JV	
2.20 Compliance with IDEAS and Public Procurement Orders	
2.21 Continued Eligibility	
2.22 Acknowledgement by the Bidder	38

2.23 Amendment of RFQ Document	. 38
SECTION-3 PREPARATION, SUBMISSION AND OPENING OF BIDS	. 39
3.1 Language	. 39
3.2 Clarifications	. 39
3.3 Cost of Bid	. 39
3.4 Multiple Bids	. 39
3.5 Authorisation to Sign the Bid	. 39
3.6 Documents comprising the Bid	. 40
3.7 Submission of Bid	. 40
3.8 Bid Due Date	. 41
3.9 Opening of Bids	. 41
3.10 Confidentiality	. 42
3.11 Correspondence with the Bidders	. 42
SECTION-4 EVALUATION OF BIDS	. 43
4.1 Evaluation Process	. 43
4.2 Tests of Responsiveness	. 44
4.3 Evaluation of Bid	. 45
4.4 Qualification of Bidders	. 45
4.5 Notification of Qualification	. 45
4.6 Subsequent Invitation for Bids	. 45
4.7 Verification and Disqualification	. 45
4.8 Rejection of Bid	. 46
SECTION-5 OTHER CONDITIONS	. 47
ART-III: APPENDIX TO INSTRUCTIONS TO BIDDERS	. 50
SECTION-6 APPENDIX TO INSTRUCTIONS TO BIDDERS	. 50
ART-IV: BIDDING FORMS AND APPENDICES	. 55
Bid Submission Form	. 56
FORM-I: Bidder Information Form	. 61
FORM-II: Pending Contract Litigation	. 65
Form-III: Contract Litigation History	
FORM-IV: Financial Status	. 69
FORM-V: Annual EPC / Construction Turnover	. 72
FORM-VI: Sources of Finance	. 74
FORM-VII: Bid Capacity	. 76
FORM-VIII: General Experience	. 78
FORM-IX: Specific Experience	
FORM-IX(A): Additional Specific Experience	
FORM-X: Certified Management System	
	SECTION-3 PREPARATION, SUBMISSION AND OPENING OF BIDS 3.1 Language 3.2 Clarifications 3.3 Cost of Bid. 3.4 Multiple Bids. 3.5 Authorisation to Sign the Bid 3.6 Documents comprising the Bid. 3.7 Submission of Bid. 3.8 Bid Due Date 3.9 Opening of Bids. 3.10 Confidentiality. 3.11 Correspondence with the Bidders SECTION-4 EVALUATION OF BIDS. 4.1 Evaluation Process 4.2 Tests of Responsiveness 4.3 Evaluation of Bid. 4.4 Qualification of Bidders. 4.5 Notification of Qualification. 4.6 Subsequent Invitation for Bids. 4.7 Verification and Disqualification. 4.8 Rejection of Bid. SECTION-5 OTHER CONDITIONS. ART-III: APPENDIX TO INSTRUCTIONS TO BIDDERS. SECTION-6 APPENDIX TO INSTRUCTIONS TO BIDDERS. BIDDING FORMS AND APPENDICES. Bid Submission Form. FORM-I: Bidder Information Form. FORM-II: Pending Contract Litigation. FORM-IV: Financial Status. FORM-V: Annual EPC / Construction Turnover. FORM-VI: Sources of Finance. FORM-VII: Bid Capacity. FORM-VIII: General Experience. FORM-VIII: General Experience. FORM-VIII: General Experience. FORM-VIII: General Experience. FORM-VIX: Specific Experience.

FORM-XI: Contractor's Equipments	84
FORM-XII(A): Skilled Human Resource	85
FORM-XII(B): Resume and Declaration	86
FORM-XIII: Information for Technical Scoring	89
FORM-XIV: Additional Requirements	90
Appendix-I: Format for Power of Attorney for signing of Bid	91
Appendix-II: Format for Authorization for Lead Member of the JV	93
Appendix-III: Format for Letter of Intent for Joint Venture	95

PART-I: SCOPE OF WORKS

The Export-Import Bank of India has approved financing of a project out of a Line of Credit extended to the *Rwanda Agriculture and Animal Resources Development Board (RAB)* for Export Targeted Modern Irrigated Agriculture Projects in Rwanda. A part of the LOC is being utilized for Establishment of Centre of Excellence in Farm Mechanisation (the "Project") and Exim Bank now invites applications from eligible Indian Entities, for completion of the Balance Works under the Project.

[A] Background of the Project

- A1. The Government of Rwanda has received a loan from the Government of India towards the cost of Export Targeting Modern Irrigated Agriculture Projects in Rwanda and intends to apply part of the funds to cover eligible payments under the Contract for the said works, with a vision to become hub of competitive and reliable technology development and transfer for Mechanized agriculture in East Africa.
- A2. The Center of Excellence (COE) is being established at Rubirizi cell, Kanombe Sector, Kicukiro district in Kigali, the capital city of Rwanda for attracting high quality researchers in the Centre of Excellence. An area of 10-12 ha is being developed under the COE with 10 Ha as demonstration farm. The site map of the Center of Excellence is as given below.



CoE in FM denotes the Centre of Excellence in Farm Mechanization

A3. Rwanda Agriculture and Animal Resources Board (RAB) under Ministry of Agriculture and Animal Resources (MINAGRI), Government of Rwanda is the "Employer" / "Project Authority" for the project. The project has been partly completed and Exim Bank now invites application for pregualification of the EPC

Contractors for completing the Balance works under the Project. The balance works shall include Engineering, Procurement, Construction, Erection, Testing and Commissioning pertaining to "Establishment of Centre of Excellence in Farm Mechanization", including all civil, structural, mechanical, electrical/electromechanical works. The project shall include annual maintenance post completion.

[B] Scope of EPC Contract

B1. General Description of Works

Undertaking balance works for establishment of Centre of Excellence in Farm Mechanization, including Engineering, Procurement, Construction, Testing and Commissioning of the following structures:

- Site Preparation & Infrastructure
 - Land clearing, leveling, and grading where required.
 - Balance works of Construction of roads, drainage, parking, and utility connections (water, electricity, sewage)
- Balance Works for Building Construction
- > Supply of Equipments & Installation
 - Farm machinery and tools (tractors, planters, harvesters)
 - Workshop equipment (welding machines, lathes, compressors)
 - Laboratory and testing equipment
 - Demonstration and training equipment
 - Energy Systems (Backup/generators and power stabilization)
- Landscaping & Environmental Management
 - Green spaces, erosion control systems, water harvesting
 - Walkways and signage.

All required materials and equipment will be procured, supplied, installed, tested and commissioned as per technical specifications /BOQ of the tender document.

B.2 Salient Features of the Centre of excellence in farm mechanization

The mission is to contribute toward the mechanization of agriculture through the generation of mechanization technologies. The COE will have the following divisions:

B.2.1 Agricultural Mechanization Research Division

Agricultural Mechanization Research Division of the Institute would be one of the technical divisions under COE to undertake and promote interdisciplinary Research and development Technology Testing & Transfer in Farm Mechanization programme in the country with following specific objectives:

- > To identify constrains and bottlenecks of agricultural mechanization in the various regions and for different crops grown by the farmers in Rwanda
- > To conduct research & development in labour saving efficient equipment & machines for mechanized agriculture
- > To conduct test, modify third country prototypes of machines, equipment &

- technologies best suited to local conditions of Rwanda
- > To test adapted implements under farmers' conditions
- > To prepare technical drawings, test reports and operators' instructions for the implements found suitable for Rwandan conditions.

The Mechanization Research Division will have following units to accomplish above objectives:

a. Pre harvest Machinery & Irrigation Research Unit

This Unit will carry out research activities and studies to develop ergonomically designed hand tools, animal operated & power operated crop production machinery for main crops like Maize, Pulses, and Cassava & Banana. Carry out adaptation trails at different locations to ascertain developed & modified agricultural machineries suitability and performance on different soil conditions, field size and moisture conditions. It will also undertake studies to reduce the initial investment cost on irrigation systems and development and design of adaptive, cost effective and appropriate irrigation systems to manage irrigation available water resources. PLC based automated testing set-ups for pumps & drippers would also be developed in the unit.

b. Postharvest Research Unit

The unit will have facilities for cleaning, grading, drying/conditioning, and milling, physico-chemical characterization and sensory evaluation. This Unit will carry out research & studies on cleaning, grading & milling (size reduction) conditioning of farm produce and food raw materials for product developments and its evaluation. The laboratory also has the facility for analysis of food raw materials and food products for their nutritional quality such as protein, carbohydrate, fat etc. The unit will also have facilities to conduct studies on packaging, handling and storage of raw materials and food products.

c. Farm Power & Energy Research Unit:

This Unit will carry out research & development and testing work on mechanical sources of farm power (diesel engines, electric motors, Tractors & Power Tillers), and Animal-machine system performances evaluation. This will also have facilities for R&D work on conversion & utilization of biomass to gas, liquid and solid fuels. The development and adoption of alternate sources of energy would also be undertaken.

d. Testing & Evaluation & Farm Unit

It will act as an important link between manufacturers and users of Agricultural machines as well as other agencies responsible for the introduction and popularization of farm equipment. Farm Machinery Testing Unit would undertake the testing of farm machinery for quality control in the manufacturing of farm machinery. This will issue test reports indicating its condition and results of adaptability to Rwandan field environment /conditions.

B.2.2 Production & Technology Transfer Division

The Production & Technology Transfer Division of COE would work in collaboration with the Research & Development Division for development of prototypes of machines suited to local conditions & would support the private participation in incountry fabrication of farming equipment/machines through prototype production. This will have following Units:

a. Prototype Production Unit

The Prototype Production Unit will exclusively be used for production of prototypes of machines/ Implements for multiplication trials, pilot introduction and also to meet the demands of users to a limited extent. This will also have modern facilities for training of manufacturers and artisans for entrepreneurship development.

b. Industrial Liaison Entrepreneurship Unit

It will develop liaison with local machinery manufacturers and will train & assist them in production of improved agricultural implements developed in the COE through R&D. This unit would promote entrepreneurship development in the country.

c. Farm Machinery Display

This would be a centrally located farm machinery display hall in the COE. It will create awareness among the local fabricators & farm machinery users through display of improved prototypes of agricultural machines & innovative technologies suited to local conditions.

B.2.3 Research Extension Division

The main aim of this division will be:

- Assessment, dissemination and commercialization of Agricultural Mechanization Technologies for enhanced productivity profitability
- Promotion of manufacturing of agricultural equipment through prototype production, capacity building and support to local manufacturers/entrepreneurs
- Capacity building, promotion of potential mechanization technologies through entrepreneurship development & public-private partnership mode
- Development of database information system on agricultural mechanizationbased technologies
- To undertake industrial extension activities to promote local production by small & medium scale manufacturers.
- > To provide training to the fabricators and end users in the fabrication and use of equipment & technology
- To coordinate extension activities through Farmers Training Centre & Sub centers of COE to familiarize the Extension Personnel of Department of Agriculture & farmers with the newly developed mechanization technologies.

a. Training Unit and Farmer Training Centre

This Unit will plan and organize short- & long-term vocational training courses in

agricultural mechanization for farmers, farm women, and rural youth with emphasis on "learning doing" for higher production. This will by help farmers/technicians/engineering student /extension workers etc. in the selection, operation, repair, maintenance, management and other aspects of mechanization. It would organize trainings for in-service field level extension workers to update their knowledge in emerging advances in agricultural mechanization research on regular basis. The Farmers Training Centre would plan & impart practice-oriented trainings on operation & maintenance of agricultural implements & machines in most scientific way using latest techniques and methodology to the practicing farmers/farm women, rural youths and field level extension functionaries for improved farming. It will also conduct off site training programmes on fuel saving techniques in tractors. Training would also be conducted on women friendly tools to farm women.

b. **Technology Promotion Unit**

This Unit would mainly carry out technology transfer activities and develop human resources for mechanization an indispensable condition for increasing the agricultural productivity and production. It organizes demonstrations, field trials and exhibitions showing the major benefits of agricultural machinery applications in the fields and giving knowledge to cut down the cost of production. It also backs up with training and communication support to the Agriculture department in their extension programmes. It would create awareness for judicious use of available machinery and power and transfer the proven technology from R& D section of COE to the actual user. It will help industry especially small sector in transplanting the research design of tested equipment/machines into industrial design for the production of quality machines.

B.2.4 Central Facilities

Following centrally located Workshops & Labs have been proposed in the COE for carrying out the Research & Development, Testing & Evaluation and Transfer of Technology in agricultural Mechanization in Rwanda & East African Region. The details of these central facilitates are given in section vi.

- Development & Assembling Unit
- > Farm Machinery Testing Unit
- Prototype Manufacturing Unit
- Machine Shop
- > CAD Cell
- Ergonomics Electric & Electronics Lab
- ▶ Lab
- Repair Workshop
- > Foundry
- Agro-processing Unit
- Quality Testing Lab and Irrigation unit
- Soil & water Testing Lab

- > Farm Machinery Display Hall
- Farmers Training Centre
- > Farm Unit
- Library & Computer Centre
- Canteen

B.3 Description of Balance Works

The EPC contract for development of the Centre of Excellence in Farm Mechanization was initially awarded by the Project Authority, however the contract was cancelled during the project execution. Due to cancellation of the contract, some of the works associated with development of the project is pending and is expected to be completed by the contractor selected through this procurement process.

B.3.1 Building No. 01 - Administrative Building (G+2) (44.70x17.80m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards including testing and commissioning; suspended ceiling; BRC Grade slab(under staircase & porch and accountant manager room and Balance in future expansion room); Flooring tiles/finish/ and granite; Internal Water Supply and Sanitary installation including fixtures, fittings, pipes, testing and commissioning; Parapet wall (bricks masonry wall with cornice); Overhead tanks with piping works; Water Proofing of top slab; fire protection and plinth protection.

B.3.2 Building 02 - Machine Display Unit (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of roofing sheets, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, floor finishing, plumbing works, fire protection and plinth protection.

B.3.3 Building-03-Prototype Manufacturing Unit. (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, floor finishing, plumbing works, fire protection and plinth protection.

B.3.4 Building-04-Development & Assembly Unit (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, casting of BRC slab, floor finishing,

plumbing works, fire protection and plinth protection.

B.3.5 Building-05-Machine shop + wood-work shop Central machine component store (40.90x20.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work has been completed), casting of BRC slab, floor finishing, plumbing works, fire protection, and plinth protection.

B.3.6 Building-06-Foundry+ Forging and Heat treatment (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.7 Building-07-Ergonomics and Electrical & Electronics Lab. (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.8 Building-08-Quality Testing Lab & Irrigation Unit (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.9 Building-09-Alternate Energy Lab. (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.10 Building-10-Agro Processing Unit. (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.11 Building-11-Farm Machinery Testing Unit (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, floor finishing, plumbing works, fire protection and plinth protection.

B.3.12 Building-12-Repair Workshop. (12.00x21.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), suspended ceiling for office, floor finishing, plumbing works, fire protection and plinth protection.

B.3.13 Building-Building-13-Scrap Yard (13.00x9.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.14 Building-14-Panel Room (8.30x5.30)

Construction of balance bricks and repair of existing ones, casting of lintel beam, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.15 Building-15-Generator Room (15.00x12.00m)

Construction of balance bricks and repair of existing ones, fixing of doors and

windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning (for electrical 10%work is considered done), casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.16 Building-16-Tractor washing (4.50x6.00m)

Excavation, laying of lean concrete, concrete works for foundation, beams, columns, shear walls and BRC slab, supply and installation of pump.

B.3.17 Building 17. Covered Tractor Parking Area. (10.00x30.00m)

Leveling and excavation of footings, laying PCC for blinding, construction of columns and beams, construction of bricks, Fixing of roof trusses and roofing sheets, Fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning, suspended ceiling for office, casting of BRC slab, floor finishing(tiles), plumbing works, fire protection and plinth protection.

B.3.18 Building-18-Public Toilet (11.20x7.00)

Construction of balance bricks and repair of existing ones, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning, suspended ceiling for office, casting of BRC slab, floor finishing(tiles), plumbing works, fire protection and plinth protection.

B.3.19 Underground Tank with Pump Room (9.475x12.10)

Preparation of Design and Drawings, Site Clearance, Excavation for Foundation and structure, Backfilling & Compaction, Loading & Carting away of surplus soil, Hardcore stone soling of 300 mm for the Foundation with Filler material, 100 mm thick blinding PCC 1:4:8 above Hardcore for Raft foundation, Supply & Installation of water p, roofing activities, Supply and Fixing of PVC / Rubber water stopper for construction joints, RCC for Raft Foundation, columns and beams, slab and stairs (All inclusive), Supply & Fixing of Rain shade over Stair well of the Pump House including supply of Structural, steel & Roofing sheets, brickwork for the side walls including Plaster & Painting, Supply & Fixing of Inserts (Nozzles, Manhole covers, Step ladders etc.) as per details provided in Drawing PL.UGT.WD.001.R0, Supply & Fixing of 400 mm wide Gratings over Drain Channel, Supply & Application of Internal Coating of RCC Reservoirs with Sika waterproofing Mortar, Supply & Application of External Coating of the Structure with Bitumen based waterproofing compound.

Supply of Pump sets including Control Panel & Pump Controllers, Construction of Foundations, Installation, Commissioning of Pump sets & Pump Controllers, Supply of Instruments including cables, Installation & Commissioning of Instruments including cables

B.3.20 Boundary Wall (640m)

Preparation of Design and Drawings Excavate trenches for the fence foundation, depth not exceeding 1.5 m deep from reduced level, 100 mm thick blinding below foundations, Stone Masonry Foundation for the Retaining wall, Loading & Carting away of the surplus soil, 525 mm average thickness stone Retaining wall of average height of 1.73 meters with cement Mortar 1:6, 300mm thick Wire cut clay bricks wall between Columns with height of 450 mm and Wire cut clay bricks Column Piers, $400 \times 400 \times 2000$ mm high ditto at 3 M centers with 400×0 mm thick concrete coping-c10 on to of walls and columns inclined to direct RW, $40 \times 40 \times 2$ mm Square Tubes (Vertical Grills 200 mm spacing) and $50 \times 50 \times 2$ mm Square Tubes (Horizontal Grills), Main Gate and Guard House.

B.3.21 Green Belt, Pavements, Passages etc.

Excavation & Levelling and earth filling, Horticulture operations including earth filling, grassing, tree plantation, shrubs and, Supply, installation, testing and commissioning of LED Street / Compound/ High mast / pathway / landscape lighting for the entire campus, External Plumbing – Sewerage, External Plumbing - Water Supply, Storm Water Drains, External firefighting, IP based CCTV system for building security comprising of PTZ/ Fixed camera, cabling, recording, display system, and hardware software support for indoors.

B.3.22 Toilet & Corridor between Building (Building no. 5, 2, 8 & 10) (5.40x7.30m) 4 nos.

Construction of balance bricks and repair of existing ones, construction of balance lintel beam, fixing of roofing sheets, fixing of doors and windows, external and internal plastering, external and internal painting, internal electrical installation including conduiting, light fixtures, fans, exhausts, AC wiring, sockets, distribution boards testing and commissioning, suspended ceiling, casting of BRC slab, floor finishing, plumbing works, fire protection and plinth protection.

B.3.23 External and Internal asphalt Road (800.833x8.05m)

Application of Layer 2, Finishing layer, laying of kerb stones and U drains, Paving for Pedestrian, Storm water channel, Rain water harvesting system and balance retaining wall.

B.3.24 Sewerage Treatment Plant

Supply, Installation and Testing of 40 KLD STP (MBR Technology) including all civil components.

B.3.25 Supply of Equipment

The supply, installation, testing and commissioning of equipment for the following units:

- a) Equipment & Tools for Tractor & Farm Machinery Testing Unit
- b) Development & Assembly Unit Pre-Harvest
- c) Alternate Energy Research

- d) Prototype Manufacturing Unit
- e) Equipment & Tools for Machine Shop
- f) Equipment proposed for Ergonomics Lab
- g) Electrical & Electronic Lab
- h) Biochemical and Food lab
- i) Irrigation Equipment Lab
- j) Post-Harvest & Agro-Processing Unit
- k) Computer Aided Design Cell
- I) Office Furniture for Admin
- m) Soil & Water Testing Lab
- n) Repair Shop
- o) Demonstration Farm Unit
- p) Central Store
- g) Utilities Facilities / Vehicles
- r) Foundry Shop

C] Contract Execution Period

The Contract shall be executed in 48 months, comprising of:

- Construction Period: 12 months including civil works and supply, installation testing & commissioning of electromechanical / Farm Mechanization equipment
- Defects Liability Period: 24 months
- Operation & Maintenance Period: 36 months (In parallel with Defects Liability Period)

PART-II: INSTRUCTIONS TO BIDDERS (ITB)

SECTION-1 INTRODUCTION

1.1 Background

- 1.1.1 The Export-Import Bank of India (hereinafter called "Exim Bank" or "the Bank"), on behalf of the Government of India ("GOI"), extends from time-to-time, concessional credit facilities (hereinafter called "Lines of Credit" or "LOC") under the Indian Development and Economic Assistance Scheme (IDEAS) to overseas governments or their nominated agencies (hereinafter called "Borrower"), for financing projects to be executed by Indian entities.
- 1.1.2 The Borrower as specified in the Appendix to Instructions to Bidders (the "AITB") has received an LOC from Exim Bank. The Borrower, through the authority (the "Authority") as specified in the AITB intends to apply the LOC / a portion of the LOC to eligible payments for undertaking works (the "Project") as specified in the AITB.
- 1.1.3 In pursuance to the above, Exim Bank invites Tech-Commercial Bids from eligible Indian entities for qualification of EPC Contractors for the Project ("Qualification Process" or the "RFQ Process"). The Estimated Cost of the contract and the estimated duration are as specified in AITB. Details pertaining to the Scope of Works, Estimated Cost and Estimated Contract Duration are only indicative in nature and may change at the time of invitation of Financial Bids.
- 1.1.4 The Authority shall undertake the Project through award of contract on Engineering, Procurement and Construction (the "EPC") mode by way of a competitive bidding process amongst the Qualified Bidders (the "Financial Bidding" or the "RFP Process"). The bidder selected through such RFP Process (the "EPC Contractor" or "Contractor") shall be responsible for designing, engineering, procurement and construction of the Project, including remedying any defects and conducting maintenance thereof. For the purpose of this document the RFQ Process and RFP Process shall be jointly referred to as the "Procurement Process".
- 1.1.5 Payment to the Contractor shall be made by Exim Bank, under the LOC, at the request of the Borrower and upon approval by Exim Bank, and will be subject, in all respects, to the terms and conditions of the credit (or financing) agreement. No party other than the Borrower shall derive any rights from the credit agreement or have any claim to the proceeds of the LOC. The payments shall be made in accordance with the agreement between the Authority and the Contractor.
- 1.2 Brief description of RFQ Process
- 1.2.1 The RFQ process shall be conducted by Exim Bank in accordance with this **RFQ Document**. The result of the Qualification Process shall be shared by Exim Bank with the

Authority, who shall be responsible for conducting the Financial Bidding amongst the Bidders Qualified by Exim Bank in accordance with this RFQ Document, for selection of the Contractor.

- 1.2.2 This RFQ Document is being issued by Exim Bank to prospective entities (the "Bidder") interested in submitting Bids (the "Bids") subject to the provisions of this RFQ Document.
- 1.2.3 This RFQ Document and any addendum thereof shall be made available by Exim Bank on its website. The RFQ Document and the addendums can be accessed using the following link: https://www.eximbankindia.in/lines-of-credit-tenders.aspx
- 1.2.4 The RFQ Document and any addenda issued subsequent to this RFQ Document, will be deemed to form part of the RFQ Document.
- 1.2.5 The RFQ Document will be available free of any cost. The Bidder shall pay Exim Bank an upfront and non-refundable processing fee (the "Processing Fee"), through NEFT transfer, as per details as specified in the AITB. The Bidder shall provide the original or copy of any document(s) generated at the time of payment of Processing Fee as evidence of payment, along with the Bid.

For avoidance of doubt, the Processing Fee will be non-refundable fee payable by the Bidder to Exim Bank.

1.2.6 Pre-Bid Meeting

- (a) A Pre-Bid Meeting shall be convened, at the designated date, time and place as specified in the AITB. A maximum of two representatives of each prospective Bidder shall be allowed to participate in the Pre-Bid Meeting. The Bidder must submit to Exim Bank, through the means as specified under Clause 1.6 [Correspondence with Exim Bank], the relevant authorizations to the representatives for participation in the Pre-Bid Meeting, at least 3 working days prior to such Pre-Bid Meeting. Any failure to submit the required authorization within the stipulated time may lead to the representatives not being allowed to attend the Pre-Bid Meeting. If so specified in the AITB, the Pre-Bid Meeting shall be conducted online in addition to the physical meeting.
- (b) At least 3 working days prior to such Pre-Bid Meeting, the Bidder shall submit in writing to Exim Bank, through the means as specified under Clause 1.6 [Correspondence with Exim Bank], any clarifications required to be sought. Submission of such requests for clarifications shall, however, not bind Exim Bank with any obligation to respond to the same and Exim Bank reserves the right to deny response to any such clarifications. These clarifications shall be responded to by Exim Bank in the manner as set out in Clause 3.2 [Clarifications].

- (c) During the course of Pre-Bid Meeting, the Bidder is free to seek any clarifications and make any suggestions for consideration by Exim Bank. Exim Bank shall endeavor to provide clarifications and such further information as it may, in its sole discretion, consider appropriate for facilitating a fair and transparent Qualification Process.
- (d) Minutes of the Pre-Bid Meeting, if applicable, including the questions or clarifications sought by the Bidders, shall be made available to all the prospective Bidders, participating in the Pre-Bid Meeting. The Minutes shall maintain standards of confidentiality as applicable to the identity of the source. Any modification to the RFQ Document that may become necessary as a result of the Pre-Bid Meeting shall be made by Exim Bank exclusively through the issue of an addendum pursuant to Clause 2.23 [Amendment of RFQ Document].
- (e) The Bidders are advised to attend the Pre-Bid Meeting, however, the same is not mandatory requirement for participation in the Qualification Process.

1.3 Schedule of RFQ Process

Exim Bank shall endeavor to adhere to the Process **as specified** in AITB.

1.4 Reporting Currency

1.4.1 All financial information to be submitted by the Bidder in the Bid shall be in the "Reporting Currency" or "RC" as specified in AITB, unless otherwise specified in the respective form.

For the purpose of conversion of currencies for reporting the financial information, the Bidder shall use the "Reference Rates of Foreign Currency published by Financial Benchmark India Private Ltd. ("FBIL")", available on https://www.fbil.org.in/ . For the historical exchange rates, which are not made available by FBIL, the reference rate published by Reserve Bank of India ("RBI") and available on https://www.rbi.org.in/scripts/ReferenceRateArchive.aspx may be used. In case a particular currency rate is not published by RBI the selling rate of such currency shall be taken from http://www.oanda.com.

The Bidder shall adhere to the following for conversion of monetary amount to the Reporting Currency:

- The equivalent amount in Reporting Currency shall be arrived at using the exchange rate as on 31st March of respective financial year, where the information for any of the past financial years is reported;
- ii. For data related to aspects like liquidity, annual residual contract value and contracts signed in the current financial year or any other information where the current status is reported, the exchange rate prevalent as on the working

day preceding the date of Notice Inviting Bids under this RFQ Document shall be used to arrive at the Reporting Currency.

Exchange rate for conversion between USD and INR is as specified in AITB.

1.5 Contents of the RFQ Document

1.5.1 The RFQ Document comprises of the disclaimer set forth at the beginning, the contents as listed below, and any Addenda issued in accordance with Clause 2.23 [Amendment of RFQ Document].

Part-I: Scope of Works

Part-II: Instructions to Bidders (ITB)

Section-1 Introduction

Section-2 Eligibility Criteria and Requirements

Section-3 Preparation, Submission and Opening of Bids

Section-4 Evaluation of Bids

Section-5 Other Conditions

Part-III: Appendix to Instruction to Bidders (AITB) Section-6 Appendix to Instructions to Bidder (AITB)

Part-IV: Bidding Forms and Appendices

- 1.5.2 In this RFQ document, Part-III: Appendix to Instructions to Bidders (AITB) shall supplement the information provided under Part-II: Instructions to Bidders (ITB) [Sections 1-5]. AITB shall be taken to modify, negate, add or alter the ITB in case of any differences between the provisions of AITB and ITB. In the event of a conflict between the ITB and AITB, the provisions of AITB shall prevail over ITB.
- 1.5.3 It is clarified that in case of conflict between the Illustrations in the document and the respective condition of the ITB / AITB, the ITB / AITB shall prevail.
- 1.5.4 Unless obtained directly from Exim Bank's Website, Exim Bank is not responsible for the completeness of the RFQ Document.
- 1.5.4 It is clarified that the "Notice inviting Bids" and the "Checklist: Forms and document to be submitted" are not part of the RFQ Document.

1.6 Correspondence with Exim Bank

Details and mode of communication for the purpose of communication, clarification, Opening of Bid is as specified in the AITB.

SECTION-2 ELIGIBILITY CRITERIA AND REQURIEMENTS

2.1 Bidders

2.1.1 A Bidder may be a single entity or a combination of entities in the form of a Joint Venture ("JV") either existing or with the intent to enter into a JV supported by a letter of intent (the "Letter of Intent" or "LOI") as per format under Sub-Clause 3.7.2(d) listing out the names of all the JV members. In the case of a JV, all members shall be jointly and severally liable for the execution of the subsequent contract(s) awarded pursuant to this Qualification Process. The JV shall be represented by an authorized representative who shall have the authority from all members of the JV so as to conduct all business for and on behalf of the JV during the Qualification Process. The JV shall comply with the terms and conditions of this RFQ Document.

2.2 Bidder's Nationality and Ownership

The Bidder shall be an Indian entity. In case of a Bidder being a JV, this Clause 2.2 [Bidder's Nationality and Ownership] shall apply individually to each member of the JV.

(a) A Bidder shall be deemed to be Indian if the Bidder is registered in India and/or incorporated/established under any law in force in India, as evidenced by its articles of incorporation and its registration documents, or any other equivalent document, as the case may be. The Bidder shall furnish such satisfactory evidence of eligibility as Exim Bank may reasonably request.

For avoidance of doubt, an entity shall be construed as a reference to any firm, company, corporation, government, state or agency of a state or any association or partnership (whether or not having separate legal personality).

(b) Any entity from:

- (i) a country which shares a land border with India (except to which the Government of India has extended any Lines of Credit or in which the Government of India is engaged in development projects) will be eligible to apply singly or as a member of JV, only if the entity is registered with the competent authority as prescribed by the Department of Promotion of Industry & Internal Trade (DPIIT), Government of India. An entity from a country which shares a land border with India means:
 - a. An entity incorporated, established or registered in such a country; or
 - A subsidiary of an entity incorporated, established or registered in such a country; or
 - c. An entity substantially controlled through entities incorporated, established or registered in such a country; or
 - d. An entity whose beneficial owner is situated in such a country;
 or
 - e. An Indian (or other) agent of such an entity; or
 - f. A natural person who is a citizen of such a country; or
 - g. A consortium or a Joint Venture where any member of the consortium or joint venture falls under any of the above.

ii) Bidder whose beneficial owner is from the country which restricts the participation of bidders from India in its own tendering, shall not be eligible to apply.

For the purpose of this Clause 2.2.1(b), the interpretation shall be in accordance with the Public Procurement Order No.4 dated February 23, 2023, issued by the Procurement Policy Division of the Department of Expenditure, Ministry of Finance, Government of India.

- 2.2.2 In case the Bidder becomes ineligible under this Clause 2.2.1, at any time post the RFQ Process (i.e. including during RFP Process and/or contract implementation phase), the Bidder shall immediately inform Exim Bank of such ineligibility condition becoming true and reasons thereof. This may lead to cancellation of Bidder's Qualification / contract.
- 2.2.3 In case of a Bidder being a JV, the conditions under this Clause 2.2 shall apply to each member of the JV.

2.2.4 Forms to be submitted

Form-I: Bidder Information Form, including Form-I(A) JV Member Information Form, if applicable, along with the attachments as outlined in the Form.

2.3 Conflict of Interest

- 2.3.1 A Bidder shall not have a conflict of interest (the "Conflict of Interest"). Any Bidder who at any point of time during the Qualification Process, or RFP Process or after award, is found to have a Conflict of Interest, shall be disqualified. In case of a Bidder being a JV, this Clause 2.3 [Conflict of Interest] shall apply individually to each member of the JV but not amongst the members of JV. A Bidder shall be deemed to have a Conflict of Interest, if:
 - i. the Bidder, its member or any constituent thereof and any other Bidder, its member or any constituent thereof directly or indirectly have common Beneficial Owner, provided that this disqualification shall not apply to any ownership by a bank, insurance company, mutual funds, pension fund or a public financial institution referred to in section 2(72) of Companies Act 2013, enacted by the Government of India. The beneficial owner shall mean:
 - a. Where the Bidder is a company, the Beneficial Owner is the natural person(s), who, whether acting alone or together, or through one or more juridical persons, has/have a controlling ownership interest or who exercise control through other means. For the purpose of this Sub-Clause 2.3.1(i)(a) "Beneficial Owner" means control over ownership of/entitlement to more than 10% (ten percent) of the shares or capital or profits of the company and "control" shall include the right to appoint majority of the directors or to control the management or policy decisions including by virtue of their shareholding or management rights or shareholders agreements or voting agreements. For avoidance of doubt, "Control" shall have the meaning given to

the term in Companies Act 2013 enacted by the Government of India.

b. Where the Bidder is a partnership firm, the Beneficial Owner is the natural person(s), who, whether acting alone or together, or through one or more juridical person, has/have control over ownership of/entitlement to more than 10% (ten percent) of capital or profits of the partnership or who exercises control through other means.

For the purpose of this Sub-Clause 2.3.1(i), "control" shall include the right to control the management or policy decision of such person, whether by operation of law or by contract or otherwise.

- ii. a member / constituent of a Bidder is also a member / constituent of another Bidder; or
- iii. a Bidder receives or has received any direct or indirect subsidy, grant, concessional loan or subordinated debt from any other Bidder, or has provided any such subsidy, grant, concessional loan or subordinated debt to any other Bidder, its member thereof. Provided that this provision shall not be applicable to a Bidder which is a Public Sector Undertaking; or
- iv. such Bidder has the same legal representative for purposes of this Bid as any other Bidder; or
- v. such Bidder has a relationship with another Bidder, directly or through common third party / parties, that puts either or both of them in a position to have access to each other's confidential information, which is not otherwise available in public domain, about, or to influence, the Bid of either or each other during the RFQ Process; or

such Bidder has participated as a consultant to the Authority in the preparation of any documents, design or technical specifications of the Project.

2.3.2 Forms to be submitted

Form-I: Bidder Information Form, including Form-I(A) JV Member Information Form, if applicable, along with the attachments as outlined in the Form.

2.4 Financial Conduct

- 2.4.1 A Bidder shall be ineligible if the Bidder or its member(s) or its promoters/directors (excluding nominee directors and independent directors):
 - i) is under default on any loan to any bank / financial institution (FI) in India and its account has been classified as Non-Performing Asset (NPA); or
 - ii) or its promoters / directors (excluding nominee directors and independent directors) are appearing in Credit Information Bureau India Ltd. (CIBIL) Defaulter List; or

- iii) has been admitted by the National Company Law Tribunal (NCLT) for initiating corporate insolvency resolution process under the Indian Bankruptcy Code (IBC); or
- iv) has been reported as fraud under the present promoter / directors by any Bank / Financial Institution in India.
- 2.4.2 In case of a Bidder being a JV, the conditions under this Clause 2.4 shall apply to each member of the JV.
- 2.4.3 In case the Bidder becomes ineligible under this Clause 2.4.1, at any time post the RFQ Process (i.e. including during RFP Process and/or contract implementation phase), the Bidder shall immediately inform Exim Bank of such ineligibility condition becoming true and reasons thereof. This may lead to cancellation of Bidder's qualification / contract.

2.4.4 Forms to be submitted

Form-I: Bidder Information Form, including Form-I(A) JV Member Information Form, if applicable, along with the attachments as outlined in the Form.

2.5 Debarment / Blacklisting or Conviction

2.5.1 The Bidder, or each member of the JV, shall not have been:

- i) debarred by Exim Bank, in accordance with Exim Bank's Debarment Policy;
- ii) blacklisted, debarred or suspended from bidding by any Multilateral Development Bank (MDB), such as World Bank, Asian Development Bank:
- iii) blacklisted, debarred or suspended from bidding by any Ministry / Authority of GOI or of the Ministry / Authority of the Government in Authority's Country;
- iv) convicted for an offense under (a) India's Prevention of Corruption Act, 1988, or (b) the Bharatiya Nyaya Sanhita, 2023 or (c) any other law for the time being in force, for causing any loss of life or property or causing a threat to public health as part of execution of a public procurement contract, or (d) for a criminal action including but not limited to cheating / fraud, shall be ineligible to participate in the Procurement Process as applicable, during such period of time as the sanctioning authority shall have determined;
- v) In regard to matters other than security and integrity of India / Authority's Country, convicted by a Court of Law or indicted or adverse orders passed by a regulatory authority which could cast a doubt on its ability to undertake the Project or which relates to a grave offence that outrages the moral sense of the community;
- vi) In regard to matters relating to security and integrity of India / Authority's Country, facing any investigation or charge-sheeted by any agency of the Government of India / Authority's Country or convicted by a Court of Law.

- 2.5.2 In case of a Bidder being a JV, the conditions under this Clause 2.5 shall apply to each member of the JV.
- 2.5.3 In case the Bidder becomes ineligible under any of the sub-heads under Clause 2.5.1, at any time during the RFQ Process, post the RFQ Process (i.e. including during RFP Process and/or contract implementation phase), the Bidder shall immediately inform Exim Bank of such ineligibility and reasons thereof. This may lead to cancellation of Bidder's Qualification / contract.

2.5.4 Forms to be submitted

Form-I: Bidder Information Form, including Form-I(A) JV Member Information Form, if applicable, along with the attachments as outlined in the Form.

2.6 Past Contract Non-Performance

- 2.6.1 The Bidder or member of the JV shall, since the date **as specified in the AITB**, and till the Bid Submission Date have no instance of:
- (a) contract¹ non-performance as a result of Bidder's / Member's default.
- (b) records of poor performance, for projects in which it is acting as the 'Contractor', including but not limited to abandoning the work, rescission of the contract for reasons which are attributable to non-performance of the Bidder, inordinate delays in completion, breach of environmental or social contractual obligations, consistent history of litigation resulting in awards against the Bidder or any of the constituents, or financial failure due to insolvency and/or bankruptcy, as evidenced by, but not limited to, imposition of a penalty by an arbitral or judicial authority or a judicial pronouncement or arbitration award against the Bidder or member. Additionally, the rescission of a contract as a part of a joint venture on account of reasons other than non-performance, such as the lead partner withdrawing would entail disqualification of the Bidder.
- 2.6.2 The Bidder shall inform Exim Bank of any such instances occurring post submission of the Bid. In case the Bidder becomes ineligible under any of the sub-heads under Clause 2.6.1, at any time post the Qualification Process (i.e. including during RFP Process and/or contract implementation phase), the Bidder shall immediately inform Exim Bank of such ineligibility and reasons thereof. This may lead to cancellation of Bidder's Qualification / contract.

¹ Non-performance, shall include all contracts where (a) non-performance was not challenged by the contractor, including through referral to the dispute resolution mechanism under the respective contract, and (b) contracts that were so challenged but fully settled against the contractor. Non-performance shall not include contracts where Employers decision was overruled by the dispute resolution mechanism. Non-performance must be based on all information on fully settled disputes or litigation, i.e. dispute or litigation that has been resolved in accordance with the dispute resolution mechanism under the respective contract.

2.6.3 In case of JV, the conditions under this Clause 2.6 [Past Contract Non-Performance] shall apply to each member of the JV.

2.6.4 Forms to be submitted Form: Bid Submission Form.

2.7 Pending Contract Litigation

2.7.1 The Bidder shall have a **positive** 'Adjusted Net Worth' determined as per the financial statements for the latest financial year **as specified in AITB**, adjusted for 'Pending Contractual Litigation Value' as considered below:

- 'Pending Contractual Litigation Value' shall be deemed to be 25% of the aggregate value of pending litigation across all contracts, net of provisions made in the latest audited financial statements for the year as specified above, which shall be deducted from the net worth to arrive at the 'Adjusted Net Worth'.
- For the purpose of this Clause, Adjusted Net Worth shall mean the Net worth as per the latest financial statements *less* [25% of Pending Contractual Litigation Value *less* provisions made in the latest audited financial statements for pending contractual litigations].
- 2.7.2 In case of JV, the conditions under this Cause 2.7 shall apply to each member of the JV.
- 2.7.3 Forms to be submitted:

Form-II: Pending Contract Litigation for the Bidder / each JV Member in case of Bidder being a JV

2.8 Contract Litigation History

- 2.8.1 The aggregate amount of awards against the Bidder, if any, since Date **as specified in the AITB** and till Bid Submission Date, shall not exceed 25% of net worth as per the latest audited financial statements.
- 2.8.2 In case of JV, the conditions under this Clause 2.8 shall apply to each member of the JV.
- 2.8.3 Forms to be submitted:

Form-III: Contract Litigation History, for the Bidder / each JV Member in case of Bidder being a JV

2.9 Financial Status and Capacity

- 2.9.1 The Bidder must, over the period of last 5 (five) financial years as **specified in the AITB**, must demonstrate meeting the following conditions:
- a) Positive net worth for each of the last 5 (five) financial years;
- b) Profit (i.e. Profit Before Tax) for at least 2 (two) out of last 5 (five) financial years; and
- c) Annual EPC / Construction Revenue in any 1 (one) of the last 5 (five) financial years of not less than the amount **as specified in AITB.**
- d) Minimum required liquid assets and / or credit facilities to be made

available² for execution of the contract shall not be less than amount **as specified in AITB**.

"Financial Year" shall mean the period starting 1st April and ending on the 31st day of March every year. Last Financial Year shall mean the period ending 31st day of March immediately preceding the date of launch of this RFQ. In case the Bidder's annual accounts for the latest Financial Year are not audited and therefore the Bidder cannot make it available, the Last Financial Year shall mean the period ending 31st day of March of the penultimate Financial Year. In such case, the Bidder shall also provide the provisional results for the latest Financial Year for which the audited accounts have not been provided.

"Net Worth" shall mean the sum of subscribed and paid-up equity and reserves, minus the revaluation reserves, miscellaneous expenditure not written off and reserves not available for distribution to equity shareholders.

Liquid assets shall include unencumbered cash and bank balances / unutilized credit facilities / lines of credit or any other financial means (independent of any contractual advance payment) sufficient to meet project cashflow requirements of ongoing contracts including the present contract, net of the other commitments of the Bidder for execution of the contract. The Bidder shall continue to comply with the requirement under Clause 2.9.1(d) after the conclusion of the RFQ Process i.e. during RFP Process and/or contract implementation phase. Failure to comply with the same may lead to cancellation of Bidder's Qualification / contract.

- 2.9.2 In the event that the latest Auditor's Report contains any qualifications, exceptions, or adverse opinions, which in the opinion of Exim Bank leads to a conclusion that a material uncertainty exists about the entity's ability to continue as a going concern, the Bidder is liable to be disqualified on the ground for not meeting the Financial Capacity. Any disqualification resulting from such qualifications shall be at the sole discretion of Exim Bank and the decision thereof shall be final and binding.
- 2.9.3 In case of JV, the following shall apply for meeting the above requirement:
 - i. Each JV member shall meet the requirement under Paragraph 2.9.1(a) and 2.9.1(b),
 - ii. For condition under Paragraph 2.9.1(c) and 2.9.1(d) above, each member shall meet at least 25%, lead member shall meet at least 40% and all members together shall meet 100% of the requirement.
- 2.9.4 Forms to be submitted:
- a. Form-IV: Financial Status, for the Bidder / each JV Member in case of Bidder being a JV, duly certified by an independent Chartered

² Exchange rate prevalent as on the working day preceding the date of Invitation for Qualification shall apply [Refer Clause 1.4 [Reporting Currency].

Accountant.

- b. Form-V: Annual EPC Turnover, for the Bidder / each JV Member in case of Bidder being a JV, duly certified by an independent Chartered Accountant.
- c. Form-VI: Sources of Finance, for the Bidder / each JV Member in case of Bidder being a JV, duly certified by an independent Chartered Accountant.

2.10 Bid Capacity

2.10.1 Aggregate annual residual value of all ongoing contracts of the Bidder as on the Bid Submission Date, shall not be more than 350% of the highest annual turnover in any one of the last five (5) years which are specified under Sub-Clause 2.9.1.

Illustration: Say the Bidder has Annual Turnover of INR 100, INR 200, INR 300, INR 250, INR 150 in last five financial years, respectively. Then the highest annual turnover over last five years shall be INR 300. In such a scenario the annual residual value of ongoing contracts should not be more than INR 1050 (INR 300*350%). For calculation of annual residual value of ongoing contracts, say the residual value of all ongoing contracts of the Bidder is INR 800 consisting of Contract 'A aggregating INR 200 (to be executed in 1 year), and Contract 'B' aggregating INR 600 (to be executed in 2 years). Therefore the annualized value of Contract 'A' shall be INR 200 (INR 200/ 1 year) and that of Contract 'B' shall be INR 300 (INR 600 / 2 years). Then the total annual residual value of the contracts is INR 500 (INR 200 + INR 300).

2.10.2 In case of Bidder being a JV, this shall apply to each member of the JV.

2.10.3 Forms to be submitted:

Form-VII: Bid Capacity for the Bidder / each JV Member in case of Bidder being a JV, duly certified by an independent Chartered Accountant.

2.11 General Experience

- 2.11.1 The Bidder shall have experience in execution of (a) Design Engineering and Construction contracts, or (b) EPC contracts; for Works as specified in the AITB in the role of prime contractor or JV member to the main client, in India that has been satisfactorily completed, since the date as specified in AITB, and till Bid Submission Date.
- 2.11.2 In case of JV, the conditions under this clause 2.11 shall apply to each member of the JV.
- 2.11.3 For the purpose of meeting eligibility under this Clause, the Bidder / each member of JV may submit a maximum 3 (three) number of Contracts. In case the number of contracts submitted is more than the number specified above, Exim Bank shall review only the first 3 (three) contracts submitted sequentially as per the serially numbered list of Contracts. The decision of Exim Bank in this regard shall be final. It is clarified that only completed contracts shall be considered for establishing eligibility under this Clause 2.11, and substantially completed contracts shall not be considered for evaluation.
- 2.11.4 In order to establish eligibility under this Clause, the following

shall also apply:

- a. Only the contracts completed by the Bidder, in the capacity as prime contractor / member of JV, to the project employer, shall be considered. In particular, the contracts of parent or subsidiary or any other affiliated company of the Bidder, who is not member of the JV, unless otherwise permitted in this RFQ Document, shall not be considered.
- b. The experience of the Bidder for execution of supplies / works / contracts in-house or for a parent / promoter / subsidiary / group concerns or for Special Purpose Vehicles (SPVs), set up by the Bidder or its group companies, shall not be considered. However, the experience of the SPV can be considered in favour of the Bidder provided it is established through documentary evidence that the SPV was established by the Bidder pursuant to a concession granted / requirement stipulated by a Project Authority who is unrelated to both the Bidder and the SPV and the Letter of Acceptance of bid / Letter of Award was issued by the Project Authority in favour of the Bidder or a JV comprising the Bidder. Further, the FAC issued by the Project Authority should indicate that the Bidder has executed the associated Works contract.
- c. In case of a Bidder being a public sector undertaking with majority shareholding by the government, experience of execution of contracts for government organizations/ enterprises/ JVs shall be considered provided that such organizations/ enterprises are not under the same administrative ministry/ department through which the government holds majority shareholding in the Bidder, unless such contracts were awarded to the Bidder through an open competitive process.
- d. Only completed contracts shall be considered for evaluation and establishing eligibility under this clause 2.11.
- e. Subcontractors' experience and resumes, if any, submitted with the Bid shall not be taken into account in determining the Bidder's compliance.

2.11.5 Forms to be submitted:

Form-VIII: General Experience, of the Bidder / each JV Member in case of a Bidder being a JV.

2.11.6 Documents to be Submitted

Final Acceptance Certificate (FAC) / Completion Certificate / Testimonial Letters issued by the Project Authority for each contract giving, inter alia, details of scope and value of work executed by the Bidder, contract start and completion dates. In case of projects executed through a Special Purpose Vehicle (SPV) under a concession agreement, FAC shall be issued by the Concession Granting Authority.

- a. In case the contracts are executed by the Bidder through an SPV under a concession agreement, either singly or in joint venture, the following additional documents/information should be submitted:
 - (i) Copy of Letter of Acceptance of the bid / grant of concession issued by the Concession Granting Authority in favour of the Bidder or a JV comprising the Bidder; and
 - (ii) Copy of Concession Agreement indicating the shareholding

structure of the SPV establishing that the SPV is wholly held by the Bidder or by a JV comprising the Bidder.

b. In case the contracts are executed by the Bidder as a member of JV, a copy of Agreement between Project Authority and JV and the JV Agreement to substantiate Individual JV member's delineated roles, responsibilities and scope and value of work.

2.12 Specific Experience

- 2.12.1 The Bidder shall demonstrate following experience by way of contracts completed side the date **as specified in the AITB** and till the Bid Submission Date:
- a. Experience in executing at least the number and value of similar (a) Design Engineering and Construction contracts, or (b) EPC contracts; for Works as specified in the AITB, in the role of prime contractor or JV member to the main client / project employer, that have been satisfactorily completed during the period specified above:
- b. Additionally, the Bidder shall demonstrate either under the above contracts or under other completed contracts, completed during the period as defined in Sub-Clause 2.12.1, any other requirement as specified in the AITB.
- 2.12.2 Specific Experience of all the Members of JV on a cumulative basis would be considered for meeting the requirements stipulated under this Clause 2.12. In addition, in case of JV, each member of JV shall demonstrate experience of at least one completed contract of value not less than 50% of the value indicated under Option I in AITB clause 2.12.1(a). For meeting the requirements stipulated in clause 2.12.1(b), the Specific Experience of all the Members of JV on a cumulative basis would be considered.
- 2.12.3 For the purpose of this Clause 2.12, for establishing eligibility by the value (amount) of completed contract, the value of the contract awarded to the Bidder shall be brought to current costing level by enhancing the actual value of work at simple rate of seven percent per annum, calculated from the date of completion of contract to the Bid Due Date, as specified in this RFQ, without considering any extensions thereof, taking base of 365 days in a Year.

Illustration: Say, the RFQ is launched with Bid Due Date of July 30, 2025, and the Bidder should have completed atleast 1 contracts of INR 120 for meeting eligibility under this clause. If the Bidder submits experience of one contract of INR 100, completed on May 1, 2020, then the number of years from contract completion to Bid Due Date shall be 5.25 years (May 1, 2020 to July 30, 2025) and the value of that contract shall be increased by INR 100 * 7% * 5.25 years = INR 36.75, and the eligible value of the contract shall be INR 100 + INR 36.75 = INR 136.75, which will be considered for evaluation.

2.12.4 For contracts awarded in currency other than INR, the equivalent INR shall be calculated based on date of completion of the contract using exchange rate in accordance with Clause 1.4.1(i).

2.12.5 It is clarified that, for ascertaining eligibility on a cumulative basis, the Contract Value (or capacity if applicable) cannot be aggregated, however, the number of contracts under Options I & II can be aggregated across the JV members.

Illustration: Say under Option II, the Bidder should have completed atleast 2 contracts of USD 100 mn each. If JV member 1 (JV1) has completed one contract of USD 100 mn and JV member 2 (JV2) has completed one contract of USD 110 mn, then number of contracts can be aggregated and the JV meets the requirement. However, if JV1 has completed one contract of USD 80 mn and JV2 has completed one contract of USD 120 mn, then the value of contracts cannot be aggregated and the JV fails to meet the requirement.

2.12.6 For the purpose of meeting eligibility under Clause 2.12.1(a), the Bidder may submit a maximum 5 (five) number of Contracts. In case the number of contracts submitted is more than the number specified above, Exim Bank shall consider only the first 5 contracts submitted sequentially as per the serially numbered list of Contracts. The decision of Exim Bank in this regard shall be final. It is clarified that only completed contracts shall be considered for evaluation and establishing eligibility under this clause 2.12 and substantially completed contracts shall not be considered for evaluation.

2.12.7 In order to establish eligibility under this Clause, the following shall also apply:

- a. Only the contracts completed by the Bidder, in the capacity as prime contractor / member of JV, to the project employer, shall be considered. In particular, the contracts of parent or subsidiary or any other affiliated company of the Bidder, who is not member of the JV, unless otherwise permitted in this RFQ Document, shall not be considered.
- b. The experience of the Bidder for execution of supplies / works / contracts in-house or for a parent / promoter / subsidiary / group concerns or for Special Purpose Vehicles (SPVs), set up by the Bidder or its group companies, shall not be considered. However, the experience of the SPV can be considered in favour of the Bidder provided it is established through documentary evidence that the SPV was established by the Bidder pursuant to a concession granted / requirement stipulated by a Project Authority who is unrelated to both the Bidder and the SPV and the Letter of Acceptance of bid / Letter of Award was issued by the Project Authority in favour of the Bidder or a JV comprising the Bidder. Further, the FAC issued by the Project Authority should indicate that the Bidder has executed the associated Works contract.
- c. In case of a Bidder being a public sector undertaking with majority shareholding by the government, experience of execution of contracts for government organizations/ enterprises/ JVs shall be considered provided that such organizations/ enterprises are not under the same administrative ministry/ department through which the government holds majority shareholding in the Bidder, unless such contracts were awarded to the Bidder through an open competitive process.

- d. Only completed contracts shall be considered for evaluation and establishing eligibility under this clause 2.11.
- e. Subcontractors' experience and resumes, if any, submitted with the Bid shall not be taken into account in determining the Bidder's compliance.

2.12.8 Forms to be submitted:

Form-IX: Specific Experience of the Bidder / each JV Member in case of Bidder being a JV.

Form-IX(A): Additional Specific Experience if sought under Clause 2.12.1(b)

2.12.9 Documents to be Submitted

Final Acceptance Certificate (FAC) / Completion Certificate / Testimonial Letters issued by the Project Authority for each contract giving, inter alia, details of scope and value of work executed by the Bidder, contract start and completion dates. In case of projects executed through a Special Purpose Vehicle (SPV) under a concession agreement, FAC shall be issued by the Concession Granting Authority.

- a. In case the contracts are executed by the Bidder through an SPV under a concession agreement, either singly or in joint venture, the following additional documents/information should be submitted:
 - (i) Copy of Letter of Acceptance of the bid / grant of concession issued by the Concession Granting Authority in favour of the Bidder or a JV comprising the Bidder; and
 - (ii) Copy of Concession Agreement indicating the shareholding structure of the SPV establishing that the SPV is wholly held by the Bidder or by a JV comprising the Bidder.
- b. In case the contracts are executed by the Bidder as a member of JV, a copy of Agreement between Project Authority and JV and the JV Agreement to substantiate Individual JV member's delineated roles, responsibilities and scope and value of work.

2.13 Quality Assurance

- 2.13.1 The Bidder shall have the ISO Accreditation / Certified Quality Management System **as specified in the AITB** valid as on the date of Bid Submission.
- 2.13.2 In case of JV, the conditions under this Clause 2.13 shall apply to each member of the JV.

2.13.3 Forms to be submitted Form-X: Certified Management System

2.14 Contractor's Equipment

- 2.14.1 The Bidder must confirm that it can avail and deploy the key tools & plants / equipment as specified in the AITB for project execution.
- 2.14.2 In case of Bidder being a JV, the eligibility shall be met by the JV on combined basis for the JV as a whole and not individually for each member of the JV.

2.14.3 Forms to be submitted

Form-XI: Contractor's Equipments

2.15 Skilled Human Resource

2.15.1 The Bidder must confirm that it can deploy the personnel for the key positions meeting the requirements as specified in the AITB.

2.15.2 Forms to be submitted Form-XII: Skilled Human Resource

2.16 Manufacturer's Authorisation

2.16.1 In case of a Bidder who offers to supply and/or install equipment in accordance with the Scope of Works outlined under Part I, that the Bidder did not manufacture or otherwise produce, the Bidder shall submit an undertaking to submit Manufacturer's Authorization(s) for all major equipment in favour of the Bidder to bid for the subsequent contract confirming warranty support, in the event of Qualification and submission of Bid by the Bidder.

2.17 Technical Scoring

2.17.1 If **specified in the AITB**, the Bidders shall be assigned Score by the Bank against the parameters as **specified in the AITB**, for ascertaining the Qualification under this clause. "**Technical Score**" for the purpose of this Clause shall mean the evaluation score assigned by the Bank.

2.17.2 In order to establish eligibility under this Clause, the Bidders shall have obtained a minimum score as specified in the AITB. For avoidance of doubt, if the Bidder fails to score minimum score under Sub-Clause 2.17.2, the Bidder will be deemed to not have met the requirement under this Clause.

2.17.3 Form to be submitted Form-XII – Information for Technical Scoring

2.18 Additional Requirement

2.18.1 If specified in AITB, the Bidder shall comply with any other additional requirement as specified in the AITB.

2.18.2 Forms to be submitted, if applicable: Form-XIII: Additional Requirement

2.19 Additional Conditions for IV

In case the Bidder is a Joint Venture, it shall meet with the following additional conditions for eligibility:

- i. The maximum number of members including the Lead Member shall not exceed 3 (three).
- ii. The lead member (the "Lead Member") of the JV shall be the one with the highest share in the JV which shall not be less than 26%.
- iii. Bid shall be submitted only in the name of the JV and not in the name of constituent member(s).
- iv. The members of the Unincorporated Joint Venture shall have signed a Letter of Intent to form a JV at the RFQ stage. The Letter of Intent, shall, *inter alia*:

- a) convey the commitment(s) of all the members that the overall project management and coordination shall be carried out exclusively by the Lead Member.
- b) convey the commitment(s) of the Lead Member in accordance with this RFQ, in case the contract is awarded to the Joint Venture:
- c) clearly outline the proposed roles and responsibilities, if any, of each member of the IV;
- d) include a statement to the effect that all members of the Joint Venture shall be liable jointly and severally for all obligations of the Contractor in relation to the contract until the obligations are performed in accordance with the Conditions of Contract.
- e) Shall in all respect be governed by and interpreted in accordance with Indian Laws.
- v. No change of JV members or change in Lead Member is allowed after the Qualification, during RFP Process or currency of existence of obligations of the Contractor in the Project, with respect to the respective JV's Bid for Qualification. The composition of the JV shall be in accordance with the conditions stipulated under the RFQ Document. The JV shall be sustained for the entire RFP Process.
- vi. The Lead Member shall have the authority to conduct all businesses for and on behalf of any and all the partners of the JV during the Qualification Process, RFP Process and, in the event the JV is awarded the Contract, during its execution. All notices/correspondence with respect to the Qualification / Bidding or the Project would be sent only to the Lead Member.

For avoidance of doubt, in the event of award of the contract to a JV, all the payments to the JV shall be made to the bank account of the JV and not to the account of constituent members.

2.20 Compliance with IDEAS and Public Procurement Orders 2.20.1 The Bidder (including each member of JV, as applicable) shall adhere to the applicable guidelines, including the IDEAS Guidelines, inter alia including the requirement of meeting minimum 75% Indian Content in the contract awarded, in accordance with the IDEAS Guidelines or such other condition as may be stipulated during the RFP Process.

Bidders are advised to refer to the GOI guidelines mentioned above which are also available at https://www.eximbankindia.in/assets/pdf/loc/IDEAS_2022_07042022.p

2.20.2 The Bidder shall comply with the provisions of Public Procurement Order No.4 dated February 23, 2023, issued by the Procurement Policy Division of the Department of Expenditure, Ministry of Finance, Government of India including any clarifications / amendments, or any other Procurement Guidelines, as may be issued by the Government of India and furnish such evidence of eligibility satisfactory to Exim Bank, as Exim Bank shall reasonably request.

2.20.3 Forms to be submitted Form: Bid Submission Form.

2.21 Continued Eligibility

The Bidder undertakes that the Bidder shall continue to comply with the terms and conditions of this RFQ Document, including the eligibility conditions, and shall provide evidence of their continued eligibility in respect of the Qualification criteria during the RFP Process. In this respect, Exim Bank retains the right to request any such evidence from the Bidder at any time during the RFP Process. After award of the contract, the Contractor shall provide evidence of their continued eligibility in respect of the Qualification criteria.

In case any of disqualification / ineligibility conditions becoming true at any time post the conclusion of the evaluation process (i.e. including during RFP Process and/or contract implementation phase), the Bidder shall immediately inform Exim Bank of such disqualification / ineligibility conditions and reasons thereof. This may lead to cancellation of the Qualification / bid / agreement, at any later stage as Exim Bank may determine in its sole discretion.

2.22 Acknowledgem ent by the Bidder

It shall be deemed that by submitting the Bid, the Bidder:

- a) is not disqualified under any disqualification criteria stipulated in the RFQ Document;
- b) met all applicable qualification requirements stipulated in the RFQ Document:
- c) acknowledged that it does not have a Conflict of Interest;
- d) made a complete and careful examination of the RFQ Document and is satisfied that the Bidder meets the eligibility criteria for Qualification under this RFQ Document;
- e) agreed to be bound by the undertakings provided by it under and in terms hereof; and
- f) read and understood the Debarment Policy of Exim Bank and acknowledges that the Bidder shall be liable for debarment in case of misrepresentations in the Bid or Bid, or non-performance under any subsequent agreements or incurring debarment under the Debarment Policy of Exim Bank.

2.23 Amendment of RFQ Document

- 2.23.1 At any time prior to the last date of submission of Bid (the "Bid Due Date"), Exim Bank may, for any reason, whether at its own initiative or in response to clarifications requested by a Bidder, modify the RFQ Document by the issuance of addenda including extension of Bid Due Date.
- 2.23.2 The addendum shall be issued in the same manner as specified under Sub-Clause 1.2.3. Any addendum thus issued shall be part of the RFQ Document.
- 2.23.3 In order to afford the Bidders a reasonable time for taking an Addendum into account, or for any other reason, Exim Bank may, in its sole discretion, extend the Bid Due Date in the RFQ Document.

<u>SECTION-3 PREPARATION, SUBMISSION AND OPENING OF BIDS</u>

3.1 Language

The Bid and all related correspondence and documents in relation to the RFQ Process shall be in English language. Supporting documents and printed literature furnished by the Bidder with the Bid may be in any other language provided that they are accompanied by translations of all the pertinent passages in the English language, duly authenticated and certified by a translator. Supporting materials, which are not translated into English, may not be considered. For the purpose of interpretation and evaluation of the Bid, the English language translation shall prevail.

3.2 Clarifications

- 3.2.1 Bidders requiring any clarification on the RFQ Document may notify Exim Bank in accordance with Clause 1.6 [Correspondence with Exim Bank] and shall clearly bear the identification/title of the Project. The Bidder should send the queries before the date specified in the schedule of RFQ Process contained in Clause 1.3 [Schedule of RFQ Process].
- 3.2.2 Exim Bank shall endeavor to respond to the queries raised or clarifications sought by the Bidders and provide such further information as it may, in its sole discretion, consider appropriate for facilitating a fair and transparent process. However, Exim Bank reserves the right not to respond to any question or provide any clarification, in its sole discretion, and nothing in this RFQ Document shall be taken or read as compelling or requiring Exim Bank to respond to any question or to provide any clarification.
- 3.2.3 Exim Bank may also on its own motion, if deemed necessary, issue interpretations and clarifications to all prospective Bidders. All clarifications and interpretations issued shall be deemed to be part of the RFQ Document. Verbal clarifications and information given by Exim Bank or its employees shall not in any way or manner be binding on Exim Bank.

3.3 Cost of Bid

The Bidders shall be responsible for all of the costs associated with the preparation of their Bid and their participation in the RFQ Process. Exim Bank shall not be responsible or in any way liable for such costs, regardless of the conduct or outcome of the RFQ Process.

3.4 Multiple Bids

No Bidder shall submit more than one Bid for Qualification. A Bidder applying individually or as a member of a JV shall not be entitled to submit another Bid either individually or as a member of any other JV, as the case may be. Submission of more than one Bids by a Bidder will result in the disqualification of all such Bids in which such a Bidder is involved.

3.5 Authorisation to Sign the Bid

The Bidder shall submit a copy of Power of Attorney as per the format at **Appendix-I**, in favor of the person duly authorized to sign on behalf of the Bidder, through an appropriate board resolution or such equivalent documentation certified by an authorized officer of the

Bidder to be authentic. Certified copy of such Board resolution or such equivalent documentation shall be submitted along with the Bid. In case the Bidder is an Unincorporated JV, each member of the JV shall submit a copy of the Power of Attorney as per the format at **Appendix-II** along with a certified copy of relevant Board resolution or such equivalent documentation and the Lead Member shall also submit a copy of Power of Attorney as per format at **Appendix-I** along with a copy of relevant Board resolution or such equivalent documentation certified by the person authorized in **Appendix-I**.

3.6 Documents comprising the Bid

- 3.6.1 The Bidder shall submit all the details as required in this RFQ Document. The Bidder shall ensure that all the details are updated as on the date of submission of the Bid.
- 3.6.2 All financial / monetary information in the Bid are required to be in currency in accordance with Clause 1.4 [Reporting Currency], unless otherwise specified in the Form.
- 3.6.3 The Bid shall be furnished by the Bidder in the formats as per Part-IV: Bidding Forms and Appendices. The Bid shall comprise of:
- 1. Evidence of payment of the Processing Fee, in accordance with Sub-Clause 1.2.5;
- 2. Bid Submission Form;
- 3. Form-I: Bidder Information Form. In case of JV, Form-I(A): JV Member Information Form, shall also be submitted for each member of the JV;
- 4. Form-II: Pending Contract Litigation
- 5. Form-III: Contract Litigation History
- 6. Form-IV: Financial Status
- 7. Form-V: Annual EPC Turnover
- 8. Form-VI: Sources of Finance
- 9. Form-VII: Bid Capacity
- 10. Form-VIII: General Experience
- 11. Form-IX: Specific Experience
- 12.Form-IX(A): Additional Specific Experience (if applicable in accordance with Clause 2.12.1(b))
- 13.Form-X: Certified Management System
- 14. Form-XI: Contractor's Equipments
- 15. Form-XII: Skilled Human Resource
- 16. Form-XIII: Information for Technical Scoring
- 17.Form-XIV: Additional Requirements (if applicable in accordance with Clause 2.18)
- 18. Appendix I: Power of Attorney to sign on behalf of the Bidder
- 19. Appendix-II: Authorisation in favour of Lead Member
- 20. Appendix-III: Letter of Intent in case of Bidder being a IV.

3.7 Submission of Bid

3.7.1 The Bids shall be submitted online through the web portal which is accessible through https://eprocure.eximbankindia.in/. Bids submitted in any other manner shall not be considered for opening and evaluation and shall be rejected. Only the Bid, authenticated using the the Digital Signature Certificate (DSC) of a person authorised for signing the Bid, as evidenced from the Power of

Attorney referred to in Clause 3.7.2(b), shall be considered.

- 3.7.2 Without prejudice to the manner of Submission of Bids defined under Sub-Clause 3.7.1, the Bidder shall submit the following documents physically at the address specified in Clause 1.6 [Correspondence with Exim Bank] and by the date as specified under Clause 1.3 [Schedule of RFQ Process]:
- a. Evidence of payment of the Processing Fee in accordance with Sub-Clause 1.2.5:
- b. Original Power of Attorney (POA) / Board Resolution for signing the Bid as per format at **Appendix-I**;
- c. Where Bidder is a JV, additionally the original Power of Attorney by each member of JV as per the format at **Appendix-II**;
- d. Where Bidder is a JV, additionally the original JV Agreement or the Letter of Intent for forming Joint Venture as per the format at Appendix-III;

Above documents shall be submitted in an envelope which shall be securely sealed in such a manner that opening and resealing cannot be achieved undetected and clearly bear the name of the project in accordance with Sub-Clause 1.1.2 and the name and address of the Bidder.

3.8 Bid Due Date

- 3.8.1 Bids must be received by Exim Bank in the manner as specified under Clause 3.7 [Submission of Bid], no later than the Bid Due Date in accordance with Clause 1.3 [Schedule of RFQ Process] or any other date or time as may be extended in the manner provided in this RFQ Document. Bids received by Exim Bank after the date and time specified under Bid Due Date shall not be eligible for consideration and shall be summarily rejected.
- 3.8.2 Exim Bank may, at its discretion, extend the deadline for the submission of Bids by amending the RFQ Document in accordance with Clause 2.23 [Amendment of RFQ Document], in which case all rights and obligations of Exim Bank and Bidder previously subject to the deadline shall thereafter be subject to the deadline as extended uniformly for all Bidders.

3.9 Opening of Bids

- 3.9.1 Exim Bank shall publicly open and read out all Bids received online, at the date, time and place in accordance with Clause 1.3 [Schedule of RFQ Process] and Clause 1.6 [Correspondence with Exim Bank], in the presence of Bidder's designated representatives who choose to attend. Provided, the processes shall be concluded in accordance with Clause 1.3 even if one or all the Bidders choose not to be represented during the opening of Bids.
- 3.9.2 The Bid of only those Bidders shall be opened whose documents have been received physically in accordance with Sub-Clause 3.7.2.
- 3.9.3 The Bidder's representatives who are present for Bid Opening shall be requested to sign the record of attendance. The omission of a Bidder's signature on the record shall not invalidate the contents

and effect of the record.

- 3.9.4 Exim Bank will subsequently examine and evaluate the Bids in accordance with the provisions set out in Section 4 [Evaluation of Bids] of RFQ Document.
- 3.9.5 The Bidders are advised that the Qualification Process will be entirely at the discretion of Exim Bank. Notwithstanding anything contained in this RFQ Document, the Exim reserves the right to accept any Bid, and to or reject any or all Bids, and to annul the Qualification Process at any time, without thereby incurring any liability to the Bidders.
- 3.9.6 If any information furnished by the Bidder is found to be incomplete, or contained in formats other than those specified herein, Exim Bank may, in its sole discretion, exclude the relevant information from evaluation.

3.10 Confidentiality

- 3.10.1 Information relating to the evaluation of Bids and outcome thereof, shall not be disclosed to other Bidders or any other persons who is not officially concerned with the process, until information on the result of the Bid evaluation is communicated to all the Bidders.
- 3.10.2 Any attempt by a Bidder to influence Exim Bank in the evaluation of the Bids may result in the rejection of its Bid.
- 3.10.3 Exim Bank / Bidder will treat all information, submitted as part of the Bid, in confidence and will require all those who have access to such material to treat the same in confidence. Exim Bank / Bidder may not divulge any such information unless it is directed to do so by any statutory entity that has the power under law to require its disclosure or is to enforce or assert any right or privilege of the statutory entity or as may be required by law or in connection with any legal process.
- 3.10.4 Without prejudice to the above, Exim Bank shall have the right to share the Bid and all documents comprising and forming part of the Bid, with the Borrower / Authority as specified under Clause 1.1.2.

3.11 Correspondence with the Bidders

- 3.11.1 Save and except as provided in this RFQ Document, Exim Bank shall not entertain any correspondence with any Bidder in relation to the acceptance or rejection of any Bid. However, Exim Bank would communicate the result of evaluation to all the Bidders who have submitted an Bid in accordance with the provisions of this RFQ Document and have not been summarily rejected. After completion of the evaluation process, Exim Bank shall also disclose the result of the Qualification process to the Borrower / Authority and on its website.
- 3.11.2 Notwithstanding the above, from the time of Bid opening to the time of declaration of results, if a Bidder wishes to contact Exim Bank on any matter related to the Qualification Process, it shall do so in writing.

SECTION-4 EVALUATION OF BIDS

4.1 Evaluation Process

- 4.1.1 Exim Bank shall open the Bids at the date and time in accordance with Clause 1.3 [Schedule of RFQ Process], at the place specified in Clause 1.6 [Correspondence with Exim Bank], and in the presence of the Bidders who chooses to attend.
- 4.1.2 After opening the Bid, Exim Bank will determine if the Bid is responsive, in accordance with Clause 4.2 [Tests of Responsiveness] (the "Responsive Bid"). The Bidder shall provide all the information sought under this RFQ Document. Exim Bank will evaluate only those Bids that are received as per the manner specified under Sub-Clause 3.6 [Document Comprising the Bid] and Sub-Clause 3.7 [Submission of Bid] in the required formats and complete in all respects. Incomplete Bid or Bid where the Hard Copies of the documents required as specified in Sub-Clause 3.7.2 are not submitted, shall be liable to rejection.
- 4.1.3 For the Responsive Bids, Exim Bank will scrutinize and evaluate the Bids against the criteria specified under this RFQ. The Responsive Bids meeting the terms and conditions stipulated in the RFQ Document and meeting the requirements under Section 2 [Eligibility Criteria and Requirements], shall be declared as "Qualified Bidder".
- 4.1.4 To facilitate evaluation of Bids, Exim Bank may, at its sole discretion, but without any obligation to do so, seek clarifications in writing from any Bidder regarding its Bid. Such clarification(s) shall be provided within the time specified by the Bank for this purpose (or, if not specified, 5 days from receiving such a request). Any request for clarification(s) and all clarification(s) in response thereto shall be in writing or in electronic mode, and no change in substance of the Bid shall be sought, offered, or permitted. Any clarification submitted by a Bidder regarding its Bid that is not in response to a request by Exim Bank shall not be considered.
- 4.1.5 Exim Bank may, at its sole discretion, but without any obligation to do so, seek shortfall information / documents only, either physically or electronically after the Bid Opening, only for historical documents which pre-existed at the time of Bid Due Date as specified in this RFQ, without considering any extensions thereof, and which have not undergone change since then and do not grant any undue advantage to the Bidder.
- 4.1.6 If a Bidder does not provide clarifications sought under Sub-Clause 4.1.4 or 4.1.5 above within the prescribed time, Exim Bank shall proceed to evaluate the Bid by construing the particulars requiring clarification to the best of its understanding, and the Bidder shall be barred from subsequently questioning such interpretation of the Bank.
- 4.1.7 Exim Bank may, at its sole discretion, waive any minor issues

such as missing pages/ attachment or illegibility in a submitted document, provided that they do not constitute any material deviation. Such 'minor' issues may be conveyed to the Bidder, with a request to the Bidder to confirm and rectify the issue. The following definitions shall apply with respect to evaluation of Bids:

- (a) "**Deviation**" is a departure from the requirements specified in the RFQ Document;
- (b) "Reservation" is the setting of limiting conditions or withholding from complete acceptance of the requirements specified in the RFQ Document; and
- (c) "Omission" is the failure to submit part or all of the information or documentation required in the RFQ Document.

A 'minor' deviation/ reservation/ omission issue may be, in the sole opinion of the Bank, if accepted, would not affect in any substantial way the scope, quality, or performance of the Works, or limit in any substantial way, the rights of Exim Bank / Authority or the obligation of the Bidder during Contract execution.

Provided that if any deviation is waived for a Bidder, Exim Bank shall waive such deviation for other Bidders.

4.2 Tests of Responsiveness

4.2.1 As a first step towards evaluation of Bids, Exim Bank shall determine whether each Bid is responsive to the requirements of this RFQ Document. A Bid shall be considered responsive only if, subject to Clause 4.1.7 above:

- (a) It is received in the manner specified in Clause 3.7 [Submission of Bid];
- (b) It is received by the Bid Due Date including any extension thereof;
- (c) It is signed and submitted in accordance with Clause 3.6 [Documents comprising the Bid];
- (d) Documents listed at Sub-Clause 3.7.2 are received physically at the address as specified in Clause 1.6 [Correspondence with Exim Bank];
- (e) It is accompanied by the evidence of payment of the Processing Fee in accordance with Sub-Clause 1.2.5
- (f) It contains all the information and documents (complete in all respects) and in formats as requested in this RFQ Document;
- (g) It contains certificates from its auditors in the formats specified in this RFQ Document; and
- (h) It does not contain any condition or qualification, unless explicitly sought by Exim Bank in accordance with this RFQ

Document.

4.2.2 Exim Bank shall reject any Bid which is non-responsive and no request for alteration, modification, substitution or withdrawal shall be entertained by Exim Bank in respect of such Bid after the Bid Due Date. Exim Bank may, at its sole discretion, accept minor deviation/reservation/omission while establishing responsiveness of the Bid in accordance with Sub-Clause 4.1.7.

4.3 Evaluation of Bid

- 4.3.1 For the Responsive Bids, the Qualification of the Bidders shall be evaluated as per the requirements stipulated under Section 2 [Eligibility Criteria and Requirements] and other terms and conditions of eligibility as specified in this RFQ Document and no other methods, criteria, or requirements shall be used. Exim Bank reserves the right to waive minor deviations from the Qualification criteria if they do not materially affect the technical capability and financial resources of a Bidder to perform the contract.
- 4.3.2 The qualification, experience and resumes of sub-contractor(s), if any, submitted by the Bidder will not be considered for evaluation of the Bid.
- 4.3.3 The evaluation of the Bidders for Qualification shall be on pass/fail basis. For evaluating eligibility under Technical Scoring, the Bank shall assign scores to various criteria specified under Clause 2.17 and the overall score assigned to the Bidder shall be evaluated on Pass / Fail basis, against the minimum score specified under Clause 2.17.2

4.4 Qualification of Bidders

All Bidders, whose Bid is responsive and substantially meets or exceeds the specified eligibility criteria under this RFQ Document, and assigned Technical Score greater than or equal to the minimum score specified under Clause 2.17.2, shall be declared as "Qualified Bidders".

4.5 Notification of Qualification

- 4.5.1 Exim Bank shall notify the result of the RFQ Process through a letter to the Qualified Bidders and to the Bidders not meeting the RFQ criteria (the "Disqualified Bidders").
- 4.5.2 Bidders that have not been Qualified may write to Exim Bank no later than 7 calendar days after the date of intimation by Exim Bank, of the result of RFQ Process, to request clarifications regarding the grounds on which they were not Qualified.
- 4.5.3 Exim Bank shall also be publicly disclosing the result of RFQ Process by way of publishing the result on its website or through any other means at the discretion of Exim Bank.

4.6 Subsequent Invitation for Bids

Exim Bank shall forward the list of Qualified Bidders to the Borrower / Authority for subsequent invitation of bids.

4.7 Verification and Disqualification

Exim Bank reserves the right to verify all statements, information and documents submitted by the Bidder in response to this RFQ Document, including cross checking of the completion certificates/testimonials

with the Project Authorities issuing such certificates. The Bidder shall, when so required by Exim Bank, make available all such information, evidence and documents as may be necessary for such verification. Any such verification or lack of such verification, by Exim Bank shall not relieve the Bidder of its obligations or liabilities hereunder nor will it affect any rights of Exim Bank thereunder.

4.8 Rejection of Bid

4.8.1 Exim Bank reserves the right to reject any Bid if it is discovered that the Bidder has made misleading or false representations in the forms, statements, affidavits, and attachments submitted as a part of the Bid. If the Bidder is a Joint Venture, then the entire Joint Venture and each Member of the Joint Venture shall be disqualified/rejected.

4.8.2 Notwithstanding anything contained in this RFQ Document, Exim Bank reserves the right to reject any or all Bids and/or to annul the RFQ Process, at any time without any liability or any obligation for such acceptance, rejection or annulment and without assigning any reason thereof.

SECTION-5 OTHER CONDITIONS

- 5.1 Exim Bank, in its sole discretion and without incurring any obligation or liability, reserves the right, at any time, to;
 - a. suspend and/ or cancel the Qualification Process and/ or amend and/ or supplement the Qualification Process or modify the dates or other terms and conditions relating thereto;
 - b. consult in writing with any Bidder in order to receive clarification or further information:
 - c. select or not to select any Bidder and/ or to consult with any Bidder in order to receive clarification or further information;
 - d. retain any information and/ or evidence submitted to Exim Bank by, on behalf of, and/ or in relation to any Bidder;
 - e. independently verify, disqualify, reject and/ or accept any and all submissions or other information and/ or evidence submitted by or on behalf of any Bidder.
- It shall be deemed that by submitting the Bid, the Bidder agrees and releases Exim Bank, its employees, and advisers, irrevocably, unconditionally, fully and finally from any and all liability for claims, losses, damages, costs, expenses or liabilities in any way related to or arising from the exercise of any rights and / or performance of any obligations hereunder and the Bidding Document, pursuant hereto, and/ or in connection with the RFQ Process, to the fullest extent permitted by Applicable Law, and waives any and all rights and/ or claims it may have in this respect, whether actual or contingent, whether present or in future.
- 5.3 Bidder as a single entity or JV will need to comply with the provisions of the Memorandum of Instructions on Project Exports and Service Exports of the Reserve Bank of India and shall submit the Post Award Approval (if applicable), issued by the authorized dealer of the Bidder. It is clarified that obtaining necessary approvals pertaining to the "Foreign Exchange Management Act,1999" (FEMA) from the Reserve Bank of India / Authorised Banks, shall be the sole responsibility of the Bidder.
- The Bidder shall observe highest standards of ethics and integrity in regard to corrupt and fraudulent/prohibited practices as set forth under the IDEAS Guidelines and Laws of India and not indulge in prohibited practices or other misdemeanors, either directly or indirectly, at any stage during the RFQ Process / RFP Process or the execution of resultant contracts.
- 5.5 The Bidder shall not, by themselves or by a third party, offer, ask or accept for itself or for a third party, financial, including agency commission or non-financial advantage or even direct or indirect

promise or any other acts, as offsets for Qualification or contract award and execution as applicable, if susceptible of being qualified as criminal unlawful acts in accordance with national laws of any party involved.

- 5.6 Without prejudice to the rights of Exim Bank, if in accordance with administrative procedures of Government of India or Exim Bank, it is determined that a Bidder in the process of submission of this Bid or in case of award, during the contract performance has directly or indirectly or through an agent, engaged or indulged in any corrupt practice, fraudulent practice, anti-competitive practice, coercive practice, undesirable practice, restrictive practice, or obstructive practice, Exim Bank shall reject the Bid, recommend rejection of Bid or proposal to award the contract or the subsequent contract performance and declare the Bidder ineligible to be awarded future contract. Such Bidder shall not be eligible to participate in any RFQ / Bidding / Procurement conducted by Exim Bank or any other project being funded by Exim Bank.
- 5.7 For the purposes of this Clause, the following terms shall have the meaning hereinafter respectively assigned to them:
 - a) "corrupt practice" means (i) the offering, giving, receiving, or soliciting, directly or indirectly, of anything of value to influence the actions of any person connected with the Qualification Process / RFP Process (for avoidance of doubt, offering of employment to, or employing, or engaging in any manner whatsoever, directly or indirectly, any official of Exim Bank / Authority who is or has been associated in any manner, directly or indirectly, with the Qualification Process / RFP Process or the LOA or has dealt with matters concerning the Agreement or arising therefrom, before or after the execution thereof, at any time prior to the expiry of one year from the date such official resigns or retires from or otherwise ceases to be in the service of the Authority / Exim Bank, shall be deemed to constitute influencing the actions of a person connected with the RFP Process); or (ii) save and except as permitted under Paragraph 2.2.1(d), engaging in any manner whatsoever, whether during the Qualification Process / RFP Process or after the issue of the LOA or after the execution of the Agreement, as the case may be, any person in respect of any matter relating to the Project or the LOA or the Agreement, who at any time has been or is a legal. financial or technical adviser of the Authority / Exim Bank in relation to any matter concerning the Project;
 - b) "fraudulent practice" means a misrepresentation or omission of facts or suppression of facts or disclosure of incomplete facts, in order to influence the Qualification Process / RFP Process;
 - c) "Anti-competitive practice" means any collusion, bid rigging or anticompetitive arrangement, or any other practice coming under the purview of The Competition Act, 2002, between two or more Bidders, that may impair the transparency, fairness and the progress of the procurement process or to establish Bid Prices at artificial, non-competitive levels;

- d) "coercive practice" means impairing or harming or threatening to impair or harm, directly or indirectly, any person or property to influence any person's participation or action in the Qualification Process / RFP Process;
- e) "undesirable practice" means (i) establishing contact with any person connected with or employed or engaged by the Authority / Exim Bank with the objective of canvassing, lobbying or in any manner influencing or attempting to influence the Qualification Process / RFP Process; or (ii) having a Conflict of Interest;
- f) "restrictive practice" means forming a cartel or arriving at any understanding or arrangement among Bidders with the objective of restricting or manipulating a full and fair competition in the Qualification Process / RFP Process; and
- g) "Obstructive practice" means any practice which materially impede the Exim Bank's / Authority's investigation into allegations of one or more of the above mentioned prohibited practices either by deliberately destroying, falsifying, altering; or by concealing of evidence material to the investigation; or by making false statements to investigators and/ or by threatening, harassing or intimidating any party to prevent it from disclosing its knowledge of matters relevant to the investigation or from pursuing the investigation; or by impeding the Exim Bank's / Authority's rights of audit or access to information.
- 5.8 In further pursuance of this policy, the Bidder shall permit and shall cause its agents (whether declared or not), Sub-contractors, sub-consultants, service providers, or suppliers and any personnel thereof, to permit Exim Bank / Government of India to inspect all accounts, records and other documents relating to any RFQ Process, Bid submission, and contract performance (in the case of award), and to have them audited by auditors appointed by Exim Bank.

PART-III: APPENDIX TO INSTRUCTIONS TO BIDDERS

SECTION-6 APPENDIX TO INSTRUCTIONS TO BIDDERS

Following Clauses, wherever appearing in the ITB, shall be taken to be negated or additional provisions to be added to, or existing provisions to be altered. Wherever these is any conflict between the provisions in the ITB and that in the AITB, the provisions contained in the AITB shall prevail.

Section-1: Introduction

Condition	Claus	Particulars			
	e	. articulars			
Borrower	1.1.2	(MINECOFIN	f Finance and Economic Planning l), Govt. of Rwanda		
Authority	1.1.2	Rwanda Agriculture and Animal Resources Development Board (RAB)			
Project	1.1.2	Balance Wo Excellence	Balance Works under Establishment of Centre of		
Estimated Cost of Contract	1.1.3	USD 6,813,	175.13		
Estimated Duration of Contract and	1.1.3	48 Months			
period of maintenance			for Construction + 12 Months of DLP post and 36 months of Maintenance Period etion		
Processing Fee	1.2.5	An upfront and non-refundable amount of INF 59,000 including Processing Fee of INR 50,000 (Indian Rupees Fifty Thousand Only) plus applicable GST (presently 18%) is payable through NEFT transfer as per details given below by each Bidder:			
	Bidder	's Name	<bidder's name=""></bidder's>		
	Name Benefic		' '		
	Name	ciary's Bank	Export-Import Bank of India		
	Accour		HO00INR11011004		
	IFSC Co		EIBI0HO0001		
	GST Nu		27AAACE2769D1ZS		
	inserte	on to be d for NEFT	, , ,		
Pre-Bid Meeting	1.2.6 (a)		eting shall be conducted through both Physical Mode.		
		Date, time and place for the Pre-Bid meeting in the physical mode and link for joining the Pre-Bid meeting in online mode shall be communicated the participants confirming the intent to participate in the Pre-Bid Meeting in accordance with the Clause 1.2.6(a)			

Schedule of RFQ Process	1.3	Exim Bank shall adhere to the schedule of RFQ Process as specified under:
Event Description		Date
Pre-Bid meeting		December 15, 2025 – 16:30 hrs (IST)
Last date for receiving queries		December 15, 2025 – 10:00 hrs (IST)
Bid Due Date and Time		January 5, 2026 – 15:00 hrs (IST)
Physical Submission of Documents		January 5, 2026 – 15:00 hrs (IST)
Opening of Bids [Date and	Time]	January 5, 2026 – 15:30 hrs (IST)

Reporting	1.4.1	Reporting Cu	ırrency - Indian R	lupees			
Currency or RC		Reporting Units - Crore					
Exchange Rate	1.4.1.i	Applicants shall use the following rates for conversion					
for Conversion		between USD and INR:					
between USD		As of	USD 1 = INR	1 INR=USD			
and INR		31-Mar-16	66.3329	0.0151			
		31-Mar-17	64.8386	0.0154			
		31-Mar-18	65.0441	0.0154			
		31-Mar-19	69.1713	0.0145			
		31-Mar-20	75.6650	0.0132			
		31-Mar-21	73.5047	0.0136			
		31-Mar-22	75.8071	0.0132			
		31-Mar-23	82.2169	0.0122			
		31-Mar-24	83.3411	0.0120			
		31-Mar-25	85.5814	0.0116			
Exchange Rate for Conversion	1.4.1.ii	Rate of Conv	ersion as on Dec	ember 04, 202	5		
between USD			JSD = 1 INR	1 INR = USD			
and INR			89.98	0.0111			
<u> </u>	1.0	Data ila af Es	'an David for Com		:'C' L'		
Correspondence with Exim Bank	1.6	Details of Exim Bank for Communication, Clarifications, Pre-Bid Meeting, Bid Due Date and Bid Opening:					
		Mr. Rohan Sharma					
		Deputy General Manager					
		Export-Import Bank of India					
			acent Ring Road				
		Kidwai Nagar [East]					
		New Delhi – 110023					
		Email: <u>infrag</u>	roup@eximbanki	<u>india.in</u>			

Section-2: Qualification Criteria and Requirements

Condition	Clause	Particulars
Period for	2.6.1	Since December 1, 2020, till Bid Due Date

ascertaining Past					
Contract Non-					
Performance					
Financial Year	2.7.1	FY 2024-25			
for ascertaining					
Positive Adjusted					
Networth					
Period for	2.8.1	Since December 1, 2020, till Bid Due Date			
ascertaining					
Contract					
Litigation History					
Period for	2.9.1	FY 2020-21 to FY 2024-2025			
ascertaining					
Financial Status					
and Capacity					
Annual EPC	2.9.1	21.05 INR Crore (Twenty-one Crore Five Lakh Indian			
Revenue	(c)	Rupees)			
Minimum	2.9.1	5.26 INR Crore (Five Crore Twenty-Six Lakh Indian			
required to	(d)	Rupees)			
Liquid Assets					
General	2.11.1	Construction of Residential / Commercial Buildings in			
Experience		India			
Cutt off Date for	2.11.1	Starting from December 01, 2018, and up to Bid Due			
ascertaining		Date			
General					
Experience	2.12.1	Charting from December 01, 2010, and up to Did Duc			
Cut off Date for	2.12.1	Starting from December 01, 2018, and up to Bid Due Date			
ascertaining Specific		Date			
Experience					
Specific	2.12.1(At least the number and value of construction			
Experience	a)	contracts for construction of residential / commercial			
		buildings as mentioned below:			
		January State Control of the Control			
		No. of Value of each			
		Contracts contract			
		(in INR Crore)			
		Option I 3 8.42			
		OR			
		Option II 2 10.53			
		OR			
		Option III 1 16.84			
Any other	2.12.1(
requirement	b)	Not Applicable			
under Specific					
Experience	2.12	(a) Overlibe Manager 1 C. d.			
Quality	2.13	(a) Quality Management System equivalent / similar			
Assurance		to ISO 9001, and			
		(b) Environmental Management System			
i .	Ī	equivalent/similar to ISO 14001			

Equipment	2.14 List of key tools & plants/ equipment for execution of the scope of work is as follows:					
		Sr. No	Equipment Type	Capacity	Number	
		1.	Compactors		2	
		2.	Excavators		2	
		3.	Grader		2	
		4.	Mixer		2	
		5.	Bulldozers		2	
		6.	Fork Lift		2	
		7.	Pickup Vehicles		2	
		8.	Diesel Generator	125 KVA	2	
		9.	Water Tank Trucks		2	
		10.	Dumping Trucks		2	
Skilled Human Resource	2.15		s of skilled human The Bidder must sub			

Position	Minimum No.	Minimum Qualification	Minimum Experience
Resident Project Manager	1	M. Tech (Civil) / B. Tech (Civil)	8 years in case of M. Tech
			10 years in case of B. Tech
Farm Machinery Specialist		B. Tech / M. Tech (Mechanical / Agriculture / or similar)	5 years in case of M.Tech
			8 years in case of B. Tech
Construction engineer [Civil/ Electrical/ Mechanical]	3 (1 Civil, 1 Electrical 1 Mechanic	Diploma (In respective branch of Engineering) / B.E/ B. Tech (In respective branch of Engineering)	15 years in case of Diploma 10 years in
	al)		case of B.E/B.Tech
Food Processing or Post Harvest Expert	1	M. Sc (In respective branch of Engineering) / B. Sc (In respective	5 years in case of M.Sc
		branch of Engineering)	10 years in case of B.Sc

Lab Technicians		1	Bachelor in Science	5 Years' experience in an Engineering Lab
Manufacturer's Authorisation	2.16	The Bid for supp a) Supp	owing shall be added to ITE der shall submit Manufactu bly of the following: ly of 50-60 hp Tractors bly of CNC Turn Mill Cer pries	urer's Authorisation
Applicability of Scoring	2.17.1	No		
Parameters for Technical scoring	2.17.1	Not App	licable	
Minimum score for Technical Qualification	2.17.2	Not App	olicable	
Additional Requirements	2.18	Not App	olicable	

PART-IV: BIDDING FORMS AND APPENDICES

BID SUBMISSION FORM

[To be filled by the Bidder / Lead Member of Bidder on behalf of the JV.]

To,

Export-Import Bank of India 7th Floor, Adjacent Ring Road Kidwai Nagar [East] New Delhi – 110023

Dear Sir,

Sub: Bid Submission and Declaration

Name of Bidder / Lead Member: Name of JV Member(s), if applicable:

- 1. With reference to your RFQ Document dated including Addendums, we, having examined the RFQ Document and understood its contents, hereby submit our Bid for the aforesaid Project. The Bid is unconditional and unqualified.
- 2. I/We acknowledge that Exim Bank will be relying on the information provided in the Bid and the documents accompanying the Bid for Qualification of the Contractor for the aforesaid Project, and we certify that all information provided in the Bid and in Forms, are true and correct; nothing has been omitted which renders such information misleading; and all documents accompanying such Bid are true copies of their respective originals.
- 3. This statement is made for the express purpose of Qualification as a Bidder for the captioned Project and Bidding for the Project thereof.
- 4. I/We confirm that I/we meet all the criteria specified in the RFQ Document and agree and undertake to abide by all the terms and conditions of the RFQ Document and I/we shall make available to Exim Bank any additional information it may find necessary or require to supplement or authenticate the Bid. I/We, however, understand that Exim Bank is not bound to seek such additional / supplemental information and may conclude its assessment of this Bid based on submissions made by me/us herewith.
- 5. I/We confirm that I/we am/are an Indian entity, as prescribed in the manner provided in the RFQ Document and I/we do not have our beneficial owners in a country which restricts the participation of bidders from India in its own tendering.
- 6. I/We agree and understand that the Bid is subject to the provisions of the RFQ Document. In no case, we shall have any claim or right of whatsoever nature if we are not Qualified for the Project or our Bid is not opened or rejected in accordance with the terms and conditions of the RFQ Document.
- 7. I/We declare that:

- a) I/We have examined and have no reservations to the RFQ Document, including any Addendum issued by Exim Bank;
- b) I/We do not have any Conflict of Interest in accordance with provisions of the RFO Document;
- c) I/We confirm that I/we am/are not under default on any loan to any bank/ financial institution (FI) and our account has not been classified as Non-Performing Asset (NPA). I/ We further confirm that none of our promoters/directors (excluding nominee directors and independent directors) are appearing in Credit Information Bureau India Ltd. (CIBIL) Defaulter List;
- d) I/we confirm that us or any of our member(s) or promoters / directors (excluding nominee directors and independent directors) have not been reported as fraud under the present promoter / directors by any Bank / Financial Institution in India;
- e) I/We confirm that we have not been debarred by Exim Bank, in accordance with Exim Bank's Debarment Policy;
- f) I/We hereby certify that we have taken steps to ensure that in conformity with the provisions of the RFQ Document, no person acting for us or on our behalf has engaged or will engage in any corrupt, fraudulent, anti-competitive, coercive, undesirable, restrictive or obstructive practices; and
- g) I/We declare that I/we have the necessary equipment and personnel / have the ability to source the necessary equipment and personnel to undertake the Project in the manner provided herein and within the timelines as may be stipulated;
- h) I/We confirm that I / we have no instance of contract non-performance as a result of our default;
- i) I/We confirm that there have not been any records of poor performance during the last five years by us, as on the date of submission of the Bid, for projects in which we are acting as the 'Contractor', including but not limited to abandoning the work, rescission of the contract for reasons which are attributable to our non-performance, inordinate delays in completion, consistent history of litigation resulting in awards against us or any of the constituents, or financial failure due to insolvency and/or bankruptcy as evidenced by but not limited to imposition of a penalty by an arbitral or judicial authority or a judicial pronouncement or arbitration award against the us or any of the constituents; and
- j) I/We confirm that I/we have no record of rescission of contract as a part of a joint venture.

8. Integrity Obligations:

- a) I/We have not directly or indirectly or through an agent engaged or indulged in any corrupt practices, fraudulent practices, anti-competitive practices, coercive practices, undesirable practices, restrictive practices or obstructive practices, as defined in the RFQ Document, in respect of any tender or request for proposal issued by or any agreement entered into with Exim Bank:
- b) I/We commit to take all measures necessary to prevent corrupt practices, unfair means and illegal activities during any stage of the Qualification or RFP Process.

- c) I/ We have not, during the RFQ Process, given, offered or promised to give, directly or indirectly or through intermediaries, any bribe, gift, consideration, reward, favour, any material or immaterial benefit or other advantage, commission, fees, brokerage or inducement to any official of Exim Bank, and/or any other intermediary involved in the RFQ Process connected directly or indirectly with the RFQ Process, or to any person, organization or third party related to the contract in exchange for any advantage in the bidding, evaluation, contracting and implementation of the Project.
- d) I/ We will not collude with other parties interested in the Project to impair the transparency, fairness and progress of the RFQ Process, RFP Process, Bid evaluation, contracting and implementation of the Project.
- e) I/ We will not accept any advantage in exchange for any corrupt practice, unfair means and illegal activities.
- 9. I/We declare that we/ any Member of the Joint Venture, or our/ its Associates are not a member of a/any other Joint Venture submitting an Bid for the Project.
- 10.I/We certify that in regard to matters other than security and integrity of the India, we/ any Member of the Joint Venture or any of our/ their Associates have not been convicted by a Court of Law or indicted or adverse orders passed by a regulatory authority which could cast a doubt on our ability to undertake the Project or which relates to a grave offence that outrages the moral sense of the community. I/We further certify that in regard to matters relating to security and integrity of the country, we/any Member of the Joint Venture or any of our/ their Associates have not been facing any investigation or charge-sheeted by any agency of the Government or convicted by a court of law.
- 11.I/We further certify that we have not been (a) blacklisted/ debarred/ sanctioned/ suspended from bidding by any Multilateral Development Banks such as World Bank, Asian Development Bank, or Ministry/Authority of Government of India (GOI) or the Authority's Country or Exim Bank, in accordance with extant rules and procedures, or (b) blacklisted, debarred or suspended from bidding by any Ministry / Authority of GOI or of the Ministry / Authority of the Government in Authority's Country (c) convicted for an offense under (i) India's Prevention of Corruption Act, 1988, or (ii) the Bharatiya Nyaya Sanhita, 2023 or (iii) any other law for the time being in force, for causing any loss of life or property or causing a threat to public health as part of execution of a public procurement contract or (iv) for a criminal action including but not limited to cheating / fraud.
- 12.I/We further certify that I/we or any member have/has not been admitted by the National Company Law Tribunal (NCLT) for initiating corporate insolvency resolution process under the Indian Bankruptcy Code (IBC).
- 13.I/We undertake that in case due to any change in facts or circumstances during the RFQ Process and/or RFP Process and/or contract execution, we are attracted by the provisions of disqualification in terms of the provisions of the RFQ Document, we shall intimate Exim Bank of the same immediately.

- 14.I/We hereby confirm that I/we do not have a close business relationship or family relationship or in its employment any near relations (defined as first blood relations, and their spouses, of the Bidder or the Bidder's spouse) in our employment any near relations of persons involved in decision making in the Project at the Authority / Exim Bank.
- 15.I/We undertake in the event that we are Qualified and invited to submit a bid, to arrange and deploy all the equipments, required to be deployed at site, in accordance with the RFQ Document.
- 16.I/We undertake in the event that we are Qualified and invited to submit a bid, to arrange and deploy key staff meeting the experience and qualification requirements, in accordance with the RFQ Document.
- 17.I/We undertake, in the event that we are Qualified and invited to submit a bid, to submit manufacturer's authorization(s) in our favour (a) to bid for the supply of equipment for the tender under reference; (b) confirming guarantee / warranty support for the equipment bid for by us; and (c) confirming that the equipment being bid for to be supplied shall be new and unused.
- 18. I/ we confirm that no legal, financial or technical adviser of the Authority / Exim Bank in relation to the Project is engaged by us.
- 19.I/We acknowledge the right of Exim Bank to reject our Bid without assigning any reason or otherwise and hereby waive, to the fullest extent permitted by Applicable Law, our right to challenge the same on any account whatsoever. I/We hereby irrevocably waive any right or remedy which we may have at any stage at law or howsoever otherwise arising to challenge or question any decision taken by Exim Bank in connection with the Qualification of the Bidder, or in connection with the RFQ Process itself, in respect of the above-mentioned Project and the terms and implementation thereof.
- 20.I/We agree to the provisions of the Public Procurement Orders No.4 dated February 23, 2023, issued by the Procurement Policy Division of the Department of Expenditure, Ministry of Finance, Government of India, and undertake to abide by the provisions of this Order. I/We have read the Clause regarding restrictions on procurement from a bidder of a country which shares a land border with India and countries that restrict the participation of Bidders / Bidders from India and on sub-contracting to contractors from such countries, and confirm that I/we do not belong to such country in accordance with the provisions of the said order dated February 23, 2024; I/We certify that I/we shall not sub-contract any work to an Agency from such countries unless such Agency is registered with the Competent Authority. I/We hereby certify that I/we fulfil all requirements in this regard and is eligible to be considered. We agree to submit/ ensure that our Sub-contractors shall submit, at Exim Bank's request, evidence of the origin of materials, equipment and services.
- 21. I/We agree to adhere to the provisions of IDEAS Guidelines, inter alia the requirement of meeting minimum percent (%) Indian Content.

- 22.I/We recognise and accept that Exim Bank is only financing the projects of the Project Authority subject to its own conditions which are set out in the funding agreement it may/ has entered into with the Project Authority / Borrower. As a matter of consequence, no legal relationship exists between Exim Bank and our company, our Joint Venture or our Sub-contractors. The Project Authority retains exclusive responsibility for the preparation and implementation of the RFP Process and the performance of the Agreement.
- 23.I/We further certify that we have not made any misleading or incorrect representations in the forms, statements, affidavits and attachments submitted as a part of the Bid.

Notes:

- a) Form to be filled by the Bidder / Lead Member of Bidder on behalf of the JV.
- b) During Online Submission, please mention N/A in the Name of JV in case the Bidder is a single Entity.
- c) Physical Document / Hard Copy of the Form is not required to be submitted or uploaded during online submission. The details to be directly entered in the online form. Only the Supporting Documents mentioned below are required to be submitted as Physical Document and copy to be uploaded.

Supporting Documents to be Uploaded:

- a) Evidence of payment of the Processing Fee
- b) Power of Attorney (POA) for signing the Bid as per the format at Appendix-I of RFQ Document or the Board Resolution in favour of signatory;
- c) In case of JV, Power of Attorney by each member of JV as per the format at Appendix-II of RFO Document;
- d) In case of JV, the JV Agreement or the Letter of Intent for forming Joint Venture as per the format at Appendix-III of the RFQ Document.

FORM-I: BIDDER INFORMATION FORM

[To be filled by Bidder if being a Single Entity or as Lead Member of JV.]

Bidder's name:
[insert full name]
In case of Joint Venture (JV), name of each member:
[insert full name of each member in JV]
Bidder's country* of registration:
[indicate country of Constitution]
Bidder's actual year of incorporation:
[indicate year of Constitution]
Bidder's legal address [registered in India]:
[insert street/ number/ town or city/ country]
Bidder's GST Number:
Bidder's PAN:
Bidder's authorized representative information:
Name: [insert full name]
Designation: [insert designation]
Address: [insert street/ number/ town or city/ country]
Telephone/Fax numbers: [insert telephone/fax numbers, including country and city codes]
E-mail address: [indicate e-mail address]
Bidder's Authorised Representative Photo ID Document:
Bidder's Authorised Representative Photo ID Number:

In case of a JV, Information regarding percentage share of each Member should be provided as per table below:

Sr. No.	Name of Member	Percentage of share in the JV
1.		
2.		

Supporting certificates / documents, for Bidder or for each member of JV in case of a JV, to be submitted as below:

- a. Copy of Memorandum and Articles of Association, if the Bidder is a body corporate, and if a partnership then a copy of its partnership deed or any other equivalent document for entities other than body corporate/ partnership
- b. Certificate of Incorporation (or equivalent documents of constitution or association), and/or documents of registration
- c. PAN and GST Registration Details
- d. Organizational chart
- e. List of Board of Directors with their complete designation in case of nominee directors
- f. The beneficial ownership with respective shareholding and nationality of shareholders
- g. A copy (self-attested on all pages) of Power of Attorney in favour of the person who has been authorised, through an appropriate Company Board Resolution or equivalent document, to sign on behalf of the Bidder
 - i. Where Bidder is a single entity, the Power of Attorney as per format at **Appendix-I**;
 - ii. A copy of relevant Company Board Resolution referred to item (g).
 - iii. Letter of intent as per format at **Appendix-III** to form JV, in case of JV.
- h. A Government issued document for photo identification of the Authorised Signatory

Note:

- a) Form to be filled by the Bidder / Lead Member of Bidder on behalf of the JV.
- b) Physical Document / Hard Copy of this form is not required to be submitted or uploaded during online submission. Only the details to be entered in this form.
- c) It is mandatory to upload the Supporting Documents along with the information submitted online.
- c) During online submission, mention "NA" for JV information in case the Bidder is a Single Entity

FORM-I(A): Details of JV Members

[The following form is in addition to Form I and applicable to the JV Members only. Details to be provided for each JV. Hard Copy not required to be submitted or uploaded. Data to be entered in the Bidding Software]

JV Member's name:

[insert full name of Bidder's JV Member]

JV Member's country* of registration:

[indicate country of registration]

IV Member's date of constitution:

[indicate date of constitution in dd/mmm/yyyy]

JV Member's legal address registered in India:

[insert street/ number/ town or city/ country]

JV Member's GST Number

JV Member's PAN Number

JV Member's authorized representative information

Name: [insert full name]

Address: [insert street/ number/ town or city/ country]

Telephone/Fax numbers: [insert telephone/fax numbers, including country and city

codes1

E-mail address: [indicate e-mail address]

Supporting certificates / documents:

- a. Copy of Memorandum and Articles of Association, if the Bidder is a body corporate, and if a partnership then a copy of its partnership deed or any other equivalent document for entities other than body corporate/ partnership
- b. Certificate of Incorporation (or equivalent documents of constitution or association), and/or documents of registration
- c. PAN and GST Registration Details
- d. Organizational chart
- e. List of Board of Directors with their complete designation in case of nominee directors
- f. The beneficial ownership with respective shareholding and nationality of shareholders
- g. A copy (self-attested on all pages) of Power of Attorney in favour of the person who has been authorised, through an appropriate Company Board Resolution or equivalent document, to sign on behalf of the Bidder
 - i. The Power of Attorney by each member of JV as per the format at Appendix-II;
 - ii. A copy of relevant Company Board Resolution referred above.

- iii. Letter of intent as per format at Appendix-III to form JV, in case of JV.
- h. A Government issued document for photo identification of the Authorised Signatory

FORM-II: PENDING CONTRACT LITIGATION

[Ref Clause 2.7]

Pending Litigation:

Are there any instances of Pending Contract Litigations: Yes / No If any Contract Litigation is Pending, provide the following details:

Bidder's Name: [insert full name]

Date: [insert day, month, year]

RFQ No. and title: [insert RFQ number and title]

Information pertaining to: [Insert name of Lead Member / Non-Lead Member as applicable]

Amount in INR Crore

Year of disput e	Amount in dispute (original currency and amount)	Contract Identification	Amount in dispute (equ INR Crore) (Exchange Rate used, if any)	Amount of specific provision already made, if any (INR Crore)
[insert financi al year]	[insert amount]	Contract Identification: [indicate complete contract name, number, date, value and any other identification]	[insert amount]	[insert amount]
		Contract Value: [Value in Original Currency]		
		Name of Employer: [insert full name]		
		Address of Employer: [insert street/city/country]		
		Matter in dispute: [indicate main issues in dispute]		
		Party who initiated the dispute: [indicate "Employer" or "Contractor"]		
		Status of dispute: [Indicate if it is being treated by the Adjudicator, under Arbitration or being dealt with by the Judiciary]		
	Total:			Total:

Notes:

- a) The Form should be filled by each member of the JV, if applicable.
- b) In case there are pending contract litigations, the aggregate value of such pending litigations should be entered online and details to be uploaded in a PDF file in the format provided in "Form: Pending Contract Litigations"
- c) Exchange rate in accordance with Clause 1.4.1(ii) of the RFQ Document to be used for conversion from Reporting Currency to Equ. INR, if required.

FORM-III: CONTRACT LITIGATION HISTORY

[Ref Clause 2.8]

Are there any instances of litigation since the date preceding five (5) years from the Bid Due Date: [Yes / No]

If Contract Litigation Awarded, provide details:

Bidder's Name: [insert full name]

Date: [insert day, month, year]

RFQ No. and title: [insert RFQ number and title]

Information pertaining to: [Insert name of Lead Member / Non-Lead Member as applicable]

Amount in INR Crore

		T		nt in INK Crore
Year of award	Award against the Bidder (in Original Currency)	Award against the Bidder (Amount in INR Crore) (along with Exchange Rate used, if any)	Contract Identification	Total Contract Amount (INR Crore)
[insert financial year]		[insert Amount]	Contract Identification: [indicate complete contract name, number, date, value and any other identification] Name of Employer: [insert full	[insert Amount]
			name] Address of Employer: [insert street/city/country]	
			Matter in dispute: [indicate main issues in dispute]	
			Party who initiated the dispute: [indicate "Employer" or "Contractor"]	
			Result of dispute: [Indicate if resolution was treated by the Adjudicator, under Arbitration or dealt with by the Judiciary and whether resolved in favour or against the Bidder]	
		Total:		

Notes:

- a) The Form should be filled by each member of the JV, if applicable.
- b) In case there is a history of litigation with Award against the Bidder, the aggregate value of such data should be entered and details to be uploaded in a PDF file as per format provided in "Form: Contract Litigation History" of the RFQ Document
- c) Exchange rate prevalent for the respective financial year, in accordance with Clause 1.4.1(i) to be used for conversion from Reporting Currency to Equ. INR, if required.

FORM-IV: FINANCIAL STATUS

[Ref Clause 2.9(a) and 2.9(b)]

[The following table shall be filled in for the Bidder and for each member of a Joint Venture and shall be certified by the independent chartered accountant. Failure to comply with this requirement may result in rejection of the Bid.]

Bidder's Name: [insert full name] Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to:[Insert name of |V leader or |V member/s]

Financial data

Amount in INR Crore

Type of Financial information	Historic information for previous five (5) <i>years</i>						
	LFY-4	LFY-3 Years	LFY-2	LFY-1	LFY		
	Years		Years	Year	[Last completed Financial Year]		
Share Holder's Fund	<u> </u>			1	1		
Loan Fund	Anr	lication of Fund					
	Арр	lication of Fund	u	-			
Fixed Assets							
Investments							
Net Current Assets							
(i) Current assets, loans and advances							
Less: (ii) Current liabilities & provisions							
Misc. exp. to the extent not W/Off or adjusted							
Profit and Loss Account							
Net Worth							
Net Worth							
Information from Income Statement							
Income							
Expenditure							
Profit/(Loss) Before Tax (PBT)							
Profit/(Loss) After Tax (PAT)							
Information From Cash Flow Statement							

Cash Flow from Operating Activities			
Cash Flow from Investments			
Cash Flow from Financing Activities			

^{*} Refer Clause 1.4 [Reporting Currency] for conversion and source of exchange rate

Note: Mention the Financial Years at the time of filling in the Form

Registration No/ Membership No:	Stamp
(of the independent chartered accountant)	
Date:	
Place:	
UDIN:	

Supporting certificates / documents, for Bidder or for each member of JV in case of a JV, to be submitted as below:

- a. Copies of duly audited complete annual accounts³ of the Bidder and of each member (in case of Joint Venture) for preceding 5 years as specified in Sub-Clause 2.9.1 which shall:
 - i. reflect the financial situation of the Bidder or in case of JV for each member, and not of an affiliated entity (such as parent company or group member);
 - ii. be independently audited or certified in accordance with local legislation;
 - iii. be complete, including all notes to the financial statements;
 - iv. correspond to accounting periods already completed and audited.
- b. Audited Annual Reports which shall be duly signed and stamped by Statutory Auditor. Notwithstanding the same, in case the Bidder is an entity where Statutory Auditor is not required to be appointed under applicable law, a certificate from an independent chartered accountant may be provided by such Bidder;

Notes:

(a) Mention the Financial Years at the time of filling in the Form

- (b) The Form should be filled for each member of the JV, if applicable,
- (c) For online submission, it is mandatory to upload the Form duly certified by an Independent Chartered Accountant

³ Provided that in case the annual accounts for the latest Accounting Year are not audited and therefore the Bidder cannot make it available, the Bidder shall provide provisional results for the latest Accounting Year. In such a case, the Bidder shall also provide the Audited Annual Reports for 5 (five) years preceding the year for which the Audited Annual Report is not being provided.

- (d) LFY means Last completed Financial Year. For instance if the Bid is being submitted in FY 2024-25, the LYF shall be FY 2023-24, LFY-1 shall be FY 2022-23, LFY-2 shall be FY 2021-22 and so on.
- (e) For online submission, in case JV details are not applicable, "0" to be entered
- (e) Uploading Supporting documents, as specified in the RFQ Document, is mandatory

FORM-V: ANNUAL EPC / CONSTRUCTION TURNOVER

[Ref: Clause 2.9(c)]

[The following table shall be filled in for the Bidder and for each member of a Joint Venture and shall be certified by the independent chartered accountant. Failure to comply with this requirement may result in rejection of the Bid.]

Bidder's Name: [insert full name]
Date: [insert day, month, year]
RFQ No. and title: [insert RFQ Number and title]
Information pertaining to:[Insert name of JV leader or JV member/s]

Year	EPC / Co	(INR Crore) Turno	Total Turnover (INR	Turnover		
	Contracts in India	Overseas Contracts	Total EPC Turnover	Crore)		
	[A]	[B]	C = [A+B]	[D]		
LFY						
LFY – 1 Year						
LFY – 2 Years						
LFY – 3 Years						
LFY – 4 Years						

Note: RC means Reporting Currency

Registration No/ Membership No: (of the independent chartered accountant)	Stamp
Date:	
Place:	
UDIN:	

- Notes:
- (a) The Form should be filled for each member of the JV, if applicable
- (b) For online submission, it is mandatory to upload the Form duly certified by an Independent Chartered Accountant
- (c) Exchange rate for the respective financial year, in accordance with Clause 1.4.1(i) to be used for conversion from INR to Reporting Currency, for amounts reported in Reporting Currency in this form

- (d) LFY means Last completed Financial Year. For instance if the Bid is being submitted in FY 2024-25, the LYF shall be FY 2023-24, LFY-1 shall be FY 2022-23, LFY-2 shall be FY 2021-22 and so on.
- (e) For online submission, in case JV information is not applicable, "NA" to be selected in Financial Year and "0" to be mentioned in the data cells.

FORM-VI: SOURCES OF FINANCE

[Ref: Clause 2.9(d)]

[Specify sources of finance to meet the cash flow requirements for contracts currently in progress]

Bidder's Name: [insert full name]
Date: [insert day, month, year]
RFQ No. and title: [insert RFQ Number and title]
Information pertaining to:[Insert name of JV leader or JV member/s]

VII.1 Details of Credit Facilities available to the Bidder

Sr. No.	Type of Facility	Sanctione Limit	ed	Utilised Limit		Unutilised Limit	
		INR Crore		INR Crore		INR Crore	
A.	Fund Based Limits						
	a.						
	b.						
	c.						
В.	Total Fund Based						
C.	Non-fund Based Limits						
	Α						
	В						
	С						
D.	Total Non -fund Based						
E.	Total Fund and Non-fund Based Limit						

Note: (a) RC means reporting currency; (b) Equivalent USD Mn represents the value of the actual sanctioned / utilised and unutilised amounts converted to USD using Exchange rate in accordance with Clause 1.4.1(ii)

VII.2 Details of Liquid Assets such as cash and bank balance, marketable securities and any other financial means which may be used meet the cash flow requirements for works currently in progress by the Bidder

Amount in INR Crore

Sr No	Type of Liquid Assets	Amount
1.		
2.		
3.		

Registration No/ Membership No:	Stamp
(of the independent chartered accountant)	
Date:	
Place:	
UDIN:	

Supporting documents / certificates for Bidder or for each member of JV in case of a JV to be submitted as below:

Statement of Consolidated Credit Facilities (sanctioned, utilised and available) from Lead Banker or respective Bank statements duly attested by the respective banks.

- (a) The Form should be filled for each member of the JV, if applicable. For online submission, only the Aggregate Amounts to be mentioned. The details to be provided in the supporting form certified by CA.
- (b) For online submission, it is mandatory to upload the Form duly certified by an Independent Chartered Accountant along with other supporting documents.
- (c) Exchange rate in accordance with Clause 1.4.1(ii) to be used for conversion from INR to Reporting Currency, for amounts reported in Reporting Currency in this form
- (d) For online submission, in case JV information is not applicable, "0" to be mentioned in the data cells.

FORM-VII: BID CAPACITY

[Ref: Clause 2.10]

[The following table shall be filled in for the Bidder and for each member of a Joint Venture] and shall be certified by the independent chartered accountant of the Bidder/each JV member. Failure to comply with this requirement may result in rejection of the Bid.]

Reporting Currency [Enter Reporting Currency in accordance with Clause

Bidder's Name: [insert full name]

Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to: [Insert name of |V leader or |V member/s]

Contract Identification	Role of Bidder	Start Date	Contrac tual Complet ion Date	Intend ed/ Revise d Compl etion Date	Residu al period for comple tion (in years)	Residual Value of Work in Currenc y of contract	value of Work (in INR Crore)	Annual Residu al value of Work (in INR Crore)
1	2	3	4	5	6		7	[7/6]
Contract Ref: Brief Title of the works: Amount of contract: [insert amount in currency, mention currency used, exchange rate and Reporting Currency*] Name & Address of Employer:	[insert "Prime Contra ctor" or "JV Memb er"]	[indic ate date as dd/ mmm/ yyyy]	[indicate date as dd/ mmm/ yyyy]	[indicat e date as dd/ mmm/ yyyy]	[Insert period in years starting from Bid Due Date. In case less than a year, value to be conside red shall be 1]		[Insert Amount including amount billed but pending payment up to deadline for submissi on of Bid]	[Insert Amount]

PART-IV: BIDDING FORMS AND APPENDICES

Registration No/ Membership No: (of the independent chartered accountant)	Stamp
Date:	
Place:	
UDIN:	

- (a) The Form should be filled for each member of the JV, if applicable
- (b) For online submission, it is mandatory to upload the Form duly certified by an Independent Chartered Accountant
- (c) Exchange rate in accordance with Clause 1.4.1(ii) to be used for conversion from INR to Reporting Currency, for amounts reported in Reporting Currency in this form
- (d) For online submission, in case JV information is not applicable, "0" to be mentioned in the data cells.
- (e) For Residual Period for completion, in case the residual time if less than 1 year, the value should be considered as 1 (one)

FORM-VIII: GENERAL EXPERIENCE

[Ref: Clause 2.11]

[The following table shall be filled in for the Bidder and in the case of a JV, for each Member]

Reporting Currency [Enter Reporting Currency in accordance with Clause ____]

Bidder's Name: [insert full name]

Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to: [Insert name of JV leader or JV member/s]

[Identify contracts that demonstrate similar experience in India. List contracts chronologically, according to their commencement (starting) dates and attach Final Acceptance Certificates.]

Amount in USD Mn

Contract Identification	Start Date	Initial Contractu al End Date	Actual Completi on Date and Reasons for delay, if any	Role of Bidder
Contract name: [insert full name]	[indicate date as	[indicate date as		[insert "Prime
Brief Description of the works performed by the Bidder: [describe works performed briefly]	date as dd/ mmm/ yyyy]	date as dd/ mmm/ yyyy]		Contractor " or "JV Member"]
Amount of contract: [insert Amount in currency, mention currency used, exchange rate and Reporting Currency equivalent*]				
Name of Employer: [indicate full name]				
Address: [indicate street/number/town or city/country]				

Supporting certificates / documents, for Bidder or for each member of JV in case of a JV, to be submitted as below:

- a. Final Acceptance Certificate/ Completion Certificate/ Testimonial Letters issued by the Project Authority for each contract submitted under this Clause, giving, inter alia, details of scope and value of work executed by the Bidder, contract start and completion dates. In case of projects executed through a Special Purpose Vehicle (SPV) under a concession agreement, FAC shall be issued by the Concession Granting Authority.
- b. In case the contracts submitted under this Clause are executed by the Bidder through an SPV under a concession agreement, either singly or in joint venture, the following additional documents/information should be submitted:
 - Copy of Letter of Acceptance of the bid/grant of concession issued by the Concession Granting Authority in favour of the Bidder or a JV comprising the Bidder; and
 - ii. Copy of Concession Agreement indicating the shareholding structure of the SPV establishing that the SPV is wholly held by the Bidder or by a JV comprising the Bidder;
- c. In case the contracts submitted under this Clause are executed by the Bidder as a member of JV, a copy of Agreement between Project Authority and JV and the JV Agreement to substantiate Individual JV member's delineated roles, responsibilities and scope and value of work.

- (a) For Online Submission Physical Document / Hard Copy of the Form is not required to be submitted or uploaded. Details to be entered in Bidding Software
- (b) Exchange rate for the respective financial year, in accordance with Clause 1.4.1(i) to be used for conversion from INR to Reporting Currency, for amounts reported in Reporting Currency in this form. The respective financial year shall mean the year in which the contract was signed.
- (c) Uploading Supporting documents, as specified in the RFQ Document, is mandatory
- (d) For online submission, minimum number of contracts required to be submitted under this form is 1 (one). Enter N/A for fields not applicable
- (e) It is mandatory to upload the supporting documents online

FORM-IX: SPECIFIC EXPERIENCE

[Ref: Clause 2.12(a)]

[The following table shall be filled in for contracts completed by the Bidder, and each member of a Joint Venture, as applicable]

Bidder's Name: [insert full name] Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to: [Insert name of |V leader or |V member/s]

Amount in USD Mn

Contract Identification	[insert contract name and number, if applicable]					
Contract date		[dd/mmi	m/yyyy]			
Completion date	Contractual: [dd/mmm/yy)	vy]	Actual: [dd/mmm/yyyy]			
Role in Contract [check the appropriate box]	Prime (Contractor	☐Member in JV ["]			
Total Contract Amount	[insert total of amount in concurrency]		[insert Exchange rate and total contract amount in Reporting Currency equivalent]*			
If member in a JV, specify participation in total Contract amount	[insert a percentage amount]	[insert total contract amount in local currency]	[insert exchange rate and total contract amount in Reporting Currency equivalent] *			
Employer Details	Employer's N	Employer's Name:				
	Address:					
	Telephone, fax number, E-mail:					
Description of the similarity	[Name the Sector/ Sub-sector] [Briefly mention the similarity in terms of sectoral characteristics and technical aspects listed in Sub-Factor D.2]					
For the above contract, mention	the following a	ttributes				
Capacity/ Physical size of key works items	[insert capacity/ physical size of key work items]					
2. Complexity	[insert description of complexity in accordance with the technical aspects mentioned under Part IV D.2]					
3. Methods/ Technology	[insert specific aspects of the methods/ technology involved in the contract]					
4. Rate of execution for key items	[insert execution rates for key items]					
5. Other Characteristics	[inser	[insert other characteristics as appropriate]				

Note: Rate implies the quantity executed in unit time e.g. cubic meters of earthwork per annum, etc.

Supporting certificates / documents, for Bidder or for each member of JV in case of a JV, to be submitted as below:

- a. Final Acceptance/ Completion Certificate/ Testimonial Letters issued by the Project Authority for each contract submitted under this Clause, giving, inter alia, details of scope and value of work executed by the Bidder, contract start and completion dates. In case of projects executed through a Special Purpose Vehicle (SPV) under a concession agreement, FAC shall be issued by the Concession Granting Authority.
- b. In case the contracts submitted under this Clause are executed by the Bidder through an SPV under a concession agreement, either singly or in joint venture, the following additional documents/information should be submitted:
 - i. Copy of Letter of Acceptance of the bid/grant of concession issued by the Concession Granting Authority in favour of the Bidder or a JV comprising the Bidder: and
 - ii. Copy of Concession Agreement indicating the shareholding structure of the SPV establishing that the SPV is wholly held by the Bidder or by a JV comprising the Bidder.
- c. In case the contracts submitted under this Clause are executed by the Bidder as a member of JV, a copy of Agreement between Project Authority and JV and the JV Agreement to substantiate Individual JV member's delineated roles, responsibilities and scope and value of work.

- (a) For Online Submission Physical Document / Hard Copy of the Form is not required to be submitted or uploaded. Details to be entered in Bidding Software
- (b) Exchange rate for the respective financial year, in accordance with Clause 1.4.1(i) to be used for conversion from the contract currency to INR to Reporting Currency, for amounts reported in Reporting Currency in this form. The respective financial year shall mean the year in which the contract was signed.
- (c) Uploading Supporting documents, as specified in the RFQ Document, is mandatory
- (d) For online submission, minimum number of contracts required to be submitted under this form is 1 (one). Enter N/A for fields not applicable
- (e) It is mandatory to upload the supporting documents online

FORM-IX(A): ADDITIONAL SPECIFIC EXPERIENCE

[Ref: Clause 2.12.1(b)]

[The following table shall be filled in only if additional specific experience is sought under Clause 2.12(b)(ii) of the RFQ Document]

Bidder's Name: [insert full name]
Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to: [Insert name of JV leader or JV member/s]

Amount in USD Mn

Additional Specific Experience Sought	Compliance

Supporting certificates / documents: Supporting document evidencing compliance / demonstrating additional experience sought under Clause 2.12(b)(ii) of the RFQ Document

FORM-X: CERTIFIED MANAGEMENT SYSTEM

[Ref: Clause 2.13]

[The following table shall be filled in for the Bidder and in the case of a JV, each Member]

Bidder's Name: [insert full name]

Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to: [Insert name of |V leader or |V member/s]

ISO Accreditation/ Certified Management System	Process/ Discipline Certified/ Inclusions	Certification Start Date	Certification Valid up to	Certifying Organization
[indicate the quality management system accreditation]	[indicate certificate identification number and mention which process or discipline has been certified and inclusions]	[dd/mmm/yyyy]	[dd/mmm/yyyy]	[indicate the name of the certifying organization and contact details]

Supporting certificates / documents, for Bidder or for each member of JV in case of a JV, to be submitted as below: Copy of the certifications mentioned above.

FORM-XI: CONTRACTOR'S EQUIPMENTS

[Ref: Clause 2.14]

[In case of JVCA, this form shall be submitted on a combined basis]

Bidder's Name: [insert full name]

Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to: [Insert name of JV leader or JV member/s]

We, [insert Bidder's full name and in case of JVCA, insert names of all members] hereby undertake, in the event that we are qualified and invited to submit a Financial Bid, to arrange and deploy following equipment, in accordance with Clause 2.14, at site:

Equipment	Make/	Capacity	Number
Туре	Model		
[mention	[indicate	[indicate	[indicate
the generic	make/	capacity of	number
name/ type	model of	equipment]	available]
of	equipment]		
equipment]			

⁽a) For Online Submission Physical Document / Hard Copy of the Form is not required to be submitted or uploaded. Details to be entered in Bidding Software

⁽b) Manufacturer's Authorisation to be uploaded, wherever applicable

FORM-XII(A): SKILLED HUMAN RESOURCE

[Ref: Clause 2.15]

[In case of JVCA, this form shall be submitted on a combined basis]

Bidder's Name: [insert full name]

Date: [insert day, month, year]

RFQ No. and title: [insert RFQ Number and title]

Information pertaining to: [Insert name of JV leader or JV member/s]

Position	Name of key staff	Academic Qualification	Specialization	Years of Experience	Experience in Countries

- (a) For Online Submission Physical Document / Hard Copy of the Form is not required to be submitted or uploaded. Details to be entered in Bidding Software
- (b) CVs of all the Staff mentioned in the Form to be uploaded in the Bidding Software. Indicative format of CV is provided as FORM-XII(B).

FORM-XII(B): Resume and Declaration

[Indicative format of CV for submission of information for human resource submitted in Form-XII(A). The following information shall be submitted for each key expert separately.]

L. General In	formation				
	and Title : [<i>Position No. a</i> eam Leader]	end Title from Form VII]			
Personnel information	Name:	Date of birth:			
	Address:	E-mail:			
	Country of Citizenship:				
Qualificatio ns	Professional qualification	Professional qualifications:			
	Academic qualifications:	Academic qualifications:			
	Specialization:				
2. Employme	nt Record				
Present Employment	Address of employer:				
	Telephone:	Contact (manager / personnel officer):			
	Fax:				
	Job title:	Years with present employer:			

3. Relevant Experience

Name of Bidder:

Summarize professional experience in reverse chronological order. Indicate particular technical and managerial experience relevant to the project. Past employment not related to the Project does not need to be included.

Project	Role	Durati involve (in Montl	ement	Relevant Experience
[Main project details]	[Role and responsibilities on the project]	From	То	[describe the experience relevant to this position]

4.	Language;	[language and	levels of	f speaking,	reading and	d writing	skills	,]
----	-----------	---------------	-----------	-------------	-------------	-----------	--------	----

Declaration

I, the undersigned [insert either "Contractor's Representative" or "Key Personnel" as applicable], certify that to the best of my knowledge and belief, the information contained herein correctly describes myself, my qualifications and my experience.

I confirm that I am available as certified in the following table and throughout the expected time schedule for this position as provided in the Bid:

Commitment	Details
Duration of appointment:	[insert the whole period (start and end dates) for which this position will be engaged]
Commitment to duration of contract:	[insert period (start and end dates) for which this Contractor's Representative or Key Personnel is available to work on this contract]

I understand that any misrepresentation or omission in this Form may:

- (a) be taken into consideration during Bid evaluation;⁴
- (b) result in my disqualification from participating in the Bid;
- (c) result in my dismissal from the contract.

Name of [Contractor's Representative or Key Personnel] : [insert name]

⁴ To be omitted in case, Experience of Key Personnel is not considered for evaluation of Bids

Signature:
Date:
Countersignature of authorized representative of the Bidder:
Signature:
Date:
Contractor's Representative and Key Personnel

FORM-XIII: INFORMATION FOR TECHNICAL SCORING

[Specify information required from Bidders for the purposes of Technical Scoring]

FORM-XIV: ADDITIONAL REQUIREMENTS [Ref: Clause 2.18]

[Additional Requirements, if any, sought under AITB Clause 2.18 to be provided here]

Appendix-I: Format for Power of Attorney for signing of Bid (Refer Sub-Clause 2.16.3)

Know a	Il men by these	presents, We	(name of the firm and
address	s of the reaistered a	iffice) do hereby i	rrevocably constitute, nominate, appoin
and aut	thorize Mr./ Ms. (na	ame),	son/daughter/wife of and
present	ly residing at	· ·	son/daughter/wife of and , who is presently employed with us and our true and lawful attorney (hereinafte name and on our behalf, all such acts
holding	the position of	, as	our true and lawful attorney (hereinafte
referre	d to as the "Attorn	ey") to do in our	name and on our behalf, all such acts
deeds a	and things as are r	necessary or requ	uired in connection with or incidental to
submis	sion of our Bid for (Qualification for t	he [insert name of the Project] including
			of Bid and other documents and writings
			s and providing information/ responses to
			pefore Exim Bank, and generally dealing
with Ex	im Bank in all matt	ers in connection	with the Project.
AND we	horoby agree to re	atify and confirm	and do hereby ratify and confirm all acts
			e by our said Attorney pursuant to and ir
			ower of Attorney and that all acts, deeds
			exercise of the powers hereby conferred
	id shall always be o		
	_		·
IN WITN	IESS WHEREOF WE	,	, THE ABOVE NAMED PRINCIPAL HAVE THIS DAY OF 2
EXECU	TED THIS POWER O	F ATTORNEY ON	THIS DAY OF 2
			For
		(Signature,	name, designation and address)
1	Witnesses:		
1.			
2.			
			(Notarized)
	Accepted		
		(Signature)	
	(Name, Title and A	ddrocc of the Att	ornov)
	Notes:	uuress or the Att	orney,
	TTOLCS.		

- Applicable stamp duty paid as per applicable law.
- The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, laid down by the Applicable Law and the charter documents of the executant(s) and when it is so required, the same should be under common seal affixed in accordance with the required procedure.

- Wherever required, the Bidder should submit for verification the extract of the charter documents and documents such as a board or shareholders' resolution/ power of attorney in favour of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Bidder.
- For a Power of Attorney executed and issued overseas, the document will also have to be legalised by the Indian Embassy and notarised in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Bidders from countries that have signed the Hague Legislation Convention 1961 are not required to be legalised by the Indian Embassy if it carries a conforming Apostille certificate.

Appendix-II: Format for Authorization for Lead Member of the JV

Whereas Exim Bank has invited Bids from interested parties for the Qualification of Bidders for [insert name of the Project] (the " Project ").
Whereas, and (collectively the "Joint Venture" constituted in accordance with the LOI) being Members of the Joint Venture constituted in accordance with the LOI are interested in applying for Qualification for the Project in accordance with the terms and conditions of the RFQ Document No dated and other connected documents in respect of the Project, and
Whereas, it is necessary for the Members of the Joint Venture constituted in accordance with the LOI to designate one of them as the Lead Member with all necessary power and authority to do for and on behalf of the Joint Venture constituted in accordance with the LOI, all acts, deeds and things as may be necessary in connection with the Joint Venture's bid for the Project and its execution.
NOW THEREFORE KNOW ALL MEN BY THESE PRESENTS
We,having our registered office atand M/shaving our registered office at, (hereinafter collectively referred to as the "Principals") do hereby irrevocably designate, nominate, constitute, appoint, having a registered office at, as the Lead Member and true and lawful attorney of the Joint Venture constituted in accordance with the LOI (hereinafter referred to as the "Authorized Entity"). We hereby irrevocably authorize the Authorized Entity (with power to sub-delegate) to conduct all business for and on behalf of the Joint Venture constituted in accordance with the LOI and any one of us during the Qualification Process and, for all or any of such acts, deeds or things as are necessary or required or incidental to the submission of its Bid for the Qualification for the Project, including but not limited to signing and submission of all Bids and other documents and writings, participate in any programs and processes prescribed under the RFQ Document, respond to queries, submit information/ documents, sign and execute contracts and undertakings consequent to acceptance of the Bid of the Joint Venture constituted in accordance with the LOI and generally to represent the Joint Venture constituted in accordance with the LOI in all its dealings with Exim Bank, and/ or any other Government Agency or any person, in all matters in connection with or relating to or arising out of the Joint Venture constituted in accordance with the LOI's Bid for Qualification for the Project and/ or any other requirements under the RFQ Document.
AND hereby agree to ratify and confirm and do hereby ratify and confirm all acts, deeds and things done or caused to be done by our said Authorized Entity pursuant to and in exercise of the powers conferred by this Authorized Entity and that all acts, deeds and things done by our said Attorney in exercise of the powers hereby conferred shall and shall always be deemed to have been done by us/ Joint Venture.
IN WITNESS WHEREOF WE THE PRINCIPALS ABOVE NAMED HAVE EXECUTED THIS POWER OF ATTORNEY ON THIS DAY OF 2
For

	(Signature)
	(Name & Title)
	For
	(Signature)
Witnesses:	
1.	
2.	
	(Executants)
(To be executed by all the Mo	embers of the Joint Venture)

- Applicable stamp duty paid as per applicable law.
- The mode of execution of the Power of Attorney should be in accordance with the procedure, if any, laid down by the applicable law and the charter documents of the executant(s) and when it is so required, the same should be under common seal affixed in accordance with the required procedure.
- Also, wherever required, the Bidder should submit for verification the extract of the charter documents and documents such as a board or shareholders' resolution/ power of attorney in favor of the person executing this Power of Attorney for the delegation of power hereunder on behalf of the Bidder.
- For a Power of Attorney executed and issued overseas, the document will also have to be legalised by the Indian Embassy and notarised in the jurisdiction where the Power of Attorney is being issued. However, the Power of Attorney provided by Bidder from countries that have signed the Hague Legislation Convention 1961 are not required to be legalised by the Indian Embassy if it carries a conforming Appostille certificate.

Appendix-III: Format for Letter of Intent for Joint Venture

(To be executed on Stamp paper of appropriate value)

	S LETTER OF INTENT (" <mark>LOI</mark> ")	is entered into on	this the	day of
AMC	DNGST			
and "Firs	{	office at nber " which expro	ession shall, unless	referred to as the
AND)			
2.	r]\0\$\$		ed under the	
"Sec	having its registered of cond Part" which exprisessors and permitted	ession shall, unles	(hereinafter s repugnant to the c	referred to as the context include its
	above mentioned parred to as the "Parties"			
WHE	EREAS,			
(A)	Exim Bank (which exp thereof, include its a (the "Bid"") by its RFO Document") for Qua "Project") through EP	administrators, suc Q Document No Alification of Bidde	cessors and assigns dated) has invited Bids (the " RFQ
(B)	The Parties are intere in accordance with the of the Project.	sted in jointly apply		
(C)	It is a necessary con provide a Letter of regarding the contemn into existence before copy of this LOI shall	Intent (LOI) sum plated formation of the Joint Bidding	nmarizing the Partie f a Joint Venture (" JV " Agreement during tl	es' understanding) which shall come
	NOW IT IS HEREBY AG 1. Definitions and			

⁵\$ Please mention name of the Lead Member of the JV/ Joint Venture, whether company, LLP, partnership, etc, stating the relevant Act (with year) under which incorporated.

⁶\$\$ Please mention name of other than Lead Member of the JV/ Joint Venture, whether company, LLP, partnership, etc, stating the relevant Act (with year) under which incorporated.

In this LOI, the capitalised terms shall, unless the context otherwise requires, have the meaning ascribed thereto under the RFQ Document.

2. Joint Venture

- The Parties do hereby agree to irrevocably constitute an unincorporated Joint Venture (the "JV" or "Joint Venture") for the purposes of jointly participating in the Qualification and thereafter in case Qualified, the RFP Process and if awarded, during the contract implementation stage for the Project (the "Process").
- 2.2 The Parties hereby undertake to participate in the Process only through this LOI signifying the intention to constitute a Joint Venture and not individually and/ or through any other Joint Venture constituted or proposed to be constituted for this Project, either directly or indirectly or through any of their associates, subsidiaries, affiliates etc.

3. Covenants

The Parties hereby undertake that in the event their Bid is declared as Qualified and is eligible to participate in the RFP Process, it shall, if required by the terms of the local laws of Project Authority's country and the conditions prescribed under the RFP Process, incorporate a joint venture / or any other form of corporate entity for entering into an agreement with the Project Authority and for performing all its obligations as the EPC Contractor in terms of the agreement for the Project.

4. Role of the Parties

The Parties hereby undertake to perform the roles and responsibilities as described below:

- a) Party of the First Part shall be the 'Lead Member' of the Joint Venture for the Process constituted in accordance with the LOI and shall have the power of attorney from all Parties for conducting all business for and on behalf of the Joint Venture constituted in accordance with the LOI during the RFQ process;
- b) The Lead Member shall exclusively carry out the overall management and coordination as and when required during the Process and all other parties shall be liable for the decisions and/or actions of the Lead Member;
- c) In case the Bidder is declared Qualified; the Lead Member conveys the commitment(s) of a Lead Member in accordance with terms and conditions in the RFQ Document;

a)	Party of the Second Part shall be _	; and {define the role of
	the Second Party or state "the other	member of the Joint Venture"};
{(0	c) Party of the Third Part shall be	^{7\$} ; and}

 7 \$ define the role of the Second Party or state "the other member of the Joint Venture"

(d)	Party of the Fourth Part shall be	^{8\$} . and]
-----	-----------------------------------	-----------------------

5. Joint and Several Liability

Notwithstanding the arrangement inter se the Parties, the Parties do hereby undertake to be jointly and severally responsible for all representations and submissions made in the Bid during the Qualification Process undertaken in accordance with the terms of the RFQ Document for the Project.

- 6. Stake in the Joint Venture constituted in accordance with the LOI
- 6.1 The Parties agree that the proportion of shareholding / participation interest/ profit share/ economic interest, amongst the Parties shall be as follows:

First Party: [minimum 26%]
Second Party:
{Third Party:}}

6.2 The Parties undertake that a minimum of 26% (twenty six per cent) of the subscribed and paid up equity share capital / participation interest / profit share / economic interest/ any such equivalent instrument representing ownership in the Bidder, as applicable, of the JV constituted in accordance with the LOI shall, at all times be held by the Lead Member.

7 Representation of the Parties

Each Party represents to the other Parties as of the date of this LOI that:

- (a) Such Party is duly organised, validly existing and in good standing under the laws of its incorporation and has all requisite power and authority to enter into this LOI:
- (b) The execution, delivery and performance by such Party of this LOI has been authorized by all necessary and appropriate corporate or governmental action and a copy of the extract of the charter documents and board resolution/ power of attorney/authorizations in favor of the person executing this LOI for the delegation of power and authority to execute this LOI on behalf of the member is annexed to this LOI, and will not, to the best of its knowledge:
 - (i) require any consent or approval not already obtained;

⁸\$ define the role of the Second Party or state "the other member of the Joint Venture"

- (ii) violate any laws presently in effect and having applicability to it;
- (iii) violate the memorandum and articles of association, by-laws or other applicable organisational documents thereof;
- (iv) violate any clearance, permit, concession, grant, license or other governmental authorization, approval, judgement, order or decree or any mortgage agreement, indenture or any other instrument to which such Party is a party or by which such Party or any of its properties or assets are bound or that is otherwise applicable to such Party; or
- (v) create or impose any liens, mortgages, pledges, claims, security interests, charges or encumbrances or obligations to create a lien, charge, pledge, security interest, encumbrances or mortgage in or on the property of such Party, except for encumbrances that would not, individually or in the aggregate, have a material adverse effect on the financial condition or prospects or business of such Party so as to prevent such Party from fulfilling its obligations under this LOI;
- (c) this LOI constitutes a legal and binding obligation of such Party, enforceable in accordance with its terms against it; and
- (d) there is no litigation pending or, to the best of such Party's knowledge, threatened to which it or any of its affiliates is a party that presently affects or which would have a material adverse effect on the financial condition or prospects or business of such Party in the fulfillment of its obligations under this LOI.

8 Termination

This LOI shall be effective from the date hereof and shall continue in full force and effect until a Joint Bidding Agreement is signed by the Members for the RFP Process. The same is contingent upon the Qualification of the JV constituted in accordance with the LOI under the requirements of the RFQ Document.

9 Miscellaneous

- 9.1 This LOI shall be governed by laws of India.
- 9.2 The Parties acknowledge and accept that this LOI shall not be amended by the Parties without the prior written consent of Exim Bank.

IN WITNESS WHEREOF THE PARTIES ABOVE NAMED HAVE EXECUTED AND DELIVERED THIS LOI AS OF THE DATE FIRST ABOVE WRITTEN.

SIGNED, SEALED AND DELIVERED SIGNED, SEALED AND DELIVERED For and on behalf of

LEAD MEMBER by: SECOND PART

(Signature) (Signature) (Name)

(Designation) (Designation) (Address)

In the presence of:

1. 2.

- 1. Applicable stamp duty paid as per applicable law.
- 2. The mode of the execution of this Letter of Intent should be in accordance with the procedure laid down by law, when it is so required, the same should be under common seal affixed in accordance with the required procedure.
- 3. Each Letter of Intent should attach a copy of the extract of the charter documents and documents such as resolution / Power of Attorney/authorizations in favour of the person executing this Letter of Intent for the delegation of power and authority to execute this Agreement on behalf of the Joint Venture member.
- 4. For a Letter of Intent executed and issued overseas, the document shall be legalised by the Indian Embassy and notarized in the jurisdiction where the Power of Attorney has been executed.